# Institutional investors' behind-the-scene monitoring and ESG disclosure

Yue Zhang<sup>1</sup> International School of Business and Finance, Sun Yat-sen University, China

November 2023

# Abstract

I investigate the effectiveness of institutional investors' monitoring through behind-the-scene activities on promoting ESG disclosure. Using difference-in-differences estimations, I document that firm ESG disclosure improves after institutional investors conduct corporate visits and discuss ESG-related issues with management, i.e. ESG-related corporate visits. This effect strengthens with visitors' demands for ESG information and weakens with firms' proprietary costs. Evidence from investors' trading behaviors suggests that the monitoring effects of these ESG-related visits are potentially exerted through threats of exit. Moreover, I document that ESG disclosure is negatively associated with divergence in ESG ratings, which could benefit the broad market participants.

Keywords: ESG disclosure, institutional investors, monitoring, soft activism, corporate visits

JEL classification: G23, G34, M14

**Declarations of interest:** None

<sup>&</sup>lt;sup>1</sup> Corresponding address: Dr. Yue Zhang, International School of Business and Finance, Sun Yat-sen University, Zhuhai Campus, Zhuhai, Guangdong, China, 519082; email: <u>zhangyue6@mail.sysu.edu.en</u>; mobile phone: +86 181-3868-5142.

Acknowledgements: This research was initiated during my research visit to the Sustainable and Green Finance Institute, National University of Singapore. I would like to thank the China Scholarship Council (CSC) for its financial support. I would like to thank Johan Sulaeman and Ania Zalewska for useful comments. I would like to thank Shuxin Zheng and Jiale Guo for their help in collecting some of the data used in this research.

## 1. Introduction

Sustainable investing has gained popularity over the past decades. As of June 2023, 5,372 institutional investors worldwide representing over \$120 trillion assets under management, have made the public commitment to incorporate environmental, social and governance (ESG) information into their investment processes (PRI, 2023). Along with this trend is the considerable gap between investors' burgeoning demand for ESG information and the supply of relevant information by firms (e.g. Gibson et al., 2020; Serafeim and Yoon, 2022; Agarwal et al., 2023). The lack of ESG disclosure possibly drives the substantial disagreement in ESG ratings which investors increasingly rely on to assess firms' ESG performance (Chatterji et al., 2016; Berg et al., 2022), and consequently hampers voluntary sustainable investing (EY, 2018; Avramov et al., 2022; Ilhan, et a., 2023; Krueger et al., 2023).

To promote ESG disclosure, various global-level and country-level initiatives have been adopted to harmonize ESG disclosure standards (Friedman et al., 2021; Krueger et al., 2023). In spite of this, given the non-financial nature of ESG information, how much and in what manners ESG issues are to be disclosed are largely left to managers' discretion (Baldini et al., 2018; Tsang et al., 2022). This calls for pressures from market participants with demands for ESG information and disciplining effects on managerial decisions to better stimulate ESG disclosure (Friedman et al., 2021). Motivated by this, this paper focuses on the role of institutional investors, perceived as a particularly powerful market force to monitor managerial decisions, in promoting ESG disclosure.

The monitoring role of institutional investors in inducing greater financial information disclosure is well established in the literature (e.g., Beyer et al., 2010; Boone and White, 2015; Bird and Karolyi, 2016; Dou et al., 2018; Abramavo et al., 2020). Moreover, recent studies also find positive effects of institutional ownership, shareholder proposals and institutional investors' public advocacy on eliciting greater ESG information (e.g. Flammer et al., 2021; Pawliczek et al.,

2021; Ilhan et al., 2023). However, many interventions from institutional investors occur behind the scenes (e.g., McCahery et al., 2016; Levit, 2019), through for instance, in-person meetings and private conversations, yet there is a dearth of empirical studies focusing on these actions. The lack of research on institutional investors' private engagements with management makes it impossible to comprehensively understand, and potentially creates biases when evaluating their monitoring effectiveness in facilitating ESG disclosure.

One important form of behind-the-scene activities through which institutional investors perform the monitoring role is corporate site visits, enabling investors to observe corporate operations, meet and discuss with management (Jiang and Yuan, 2018; Cao et al., 2022; Agarwal et al., 2023). Retrieving data regarding corporate visits is challenging since such activities are not required to be disclosed in most markets around the world. Shenzhen Stock Exchange (SZSE) in China is one exception, which mandates all firms listed on the exchange disclose detailed information for every corporate visit since 2009. Given that the meeting minutes are also available for every corporate visit, I am able to identify the visits during which institutional investors discuss ESG issues, i.e. ESG-related corporate visits, which directly capture their specific demand for ESG information. Therefore, the Chinese SZSE market provides a unique setting to evaluate the effectiveness of institutional investors' behind-the-scene monitoring on eliciting ESG disclosure.

Exploiting the unique datasets, I ask the following questions: (i) do firms respond to institutional investors' demands for ESG information (conveyed during ESG-related visits) by greater ESG disclosure, (ii) do firms respond more actively when institutional investors have stronger demands for ESG information and when ESG disclosure is associated with lower proprietary costs, (iii) what is the underlying mechanism through which these ESG-related visits exert effects on managerial decisions on ESG disclosure, and (iv) conditional on institutional investors' monitoring through these corporate visits being effective, does improvement in ESG

disclosure bring up any benefits to the market participants at large?

To answer these questions, I compile a sample of 698 SZSE-listed firms for which Bloomberg ESG disclosure scores are available covering the period between 2013 and 2020. I first document a positive relationship between a firm hosting ESG-related corporate visits by institutional investors and its subsequent ESG disclosure. To address potential endogeneity concerns, I perform difference-in-differences (DiD) estimation based on a matched sample of firms selected by nearest-neighbor matching (NNM), and document that the firms that host ESG-related visits experience greater improvement in ESG disclosure following these visits in comparison with the comparable firms that never host ESG-related visits. Moreover, the main finding still preserves after Heckman two-stage correction. In comparison, no statistically significant relationship is found between a firm hosting non-ESG-related corporate visits (i.e. the visits during which institutional investors do not discuss ESG issues with firms) and its subsequent ESG disclosure. The evidence highlights the importance of discussions of ESG issues (relative to non-ESG-related topics) in raising visited firms' awareness of ESG disclosure, confirming the effectiveness of institutional investors' soft activism in promoting ESG disclosure.

Second, I find that the positive link between ESG-related corporate visits and ESG disclosure is more prominent when a firm hosts more ESG-related visits, when institutional investors propose more ESG-related questions to firms, when the ESG-related visits are initiated by investors headquartered in geographically distant cities and by more ESG-aware investors (i.e. signatories of the UNPRI) and when only the environment-pillar which has least public disclosure in comparison with the social- and governance-pillar is assessed. Evidence also indicates that the positive link weakens for firms operating in more concentrated industries which in general are associated with higher proprietary costs (e.g., Botosan and Standord, 2005; Lang and Sul, 2014). The evidence implies that institutional investors potentially exert more intense monitoring on ESG disclosure when they have stronger ESG information demands, yet the effectiveness of their monitoring can be diluted by the reluctance of firms to disclose (value-relevant) ESG information in presence of high proprietary costs.

Third, I explore the mechanism through which ESG-related corporate visits affect firm ESG disclosure. By constructing a subsample of institutional investors – mutual funds, and assessing their trading behaviors, I document that visiting funds exhibit stronger preferences to firms performing well in ESG disclosure in comparison with the funds that did not conduct ESG-related visits before trading. The evidence indicates that institutional investors may *ex ante* discipline managerial behaviors by performing credible exit threats *ex post*, as managers have incentives to make efforts to avoid share selling by institutional investors (due to dissatisfaction with firm ESG disclosure) which may lead to stock price decline.

To understand whether the effective monitoring of institutional investors in promoting ESG disclosure, if any, brings about spillover benefits to the sustainable investing market at large, I examine the impact of ESG disclosure on ESG rating divergence. Employing the ESG ratings collected from four main rating agencies for Chinese firms, I find consistent evidence that greater ESG disclosure is associated with reduction in ESG rating divergence. The evidence speaks to the argument that firms' ESG disclosure provides a foundation of reliable and consistent ESG ratings (Berg et al., 2022).

The paper makes three important contributions to the literature. First, to the best of my knowledge, this is the first study to provide direct large-scale evidence that institutional investors' behind-the-scene monitoring activities elicit greater ESG disclosure. In comparison with the extensive evidence regarding institutional investors' impacts on improving financial information disclosure (Beyer et al., 2010; Boone and White, 2015; Bird and Karolyi, 2016; Dou et al., 2018; Abramavo et al., 2020), less attention is paid to their monitoring role in ESG disclosure. While

several recent papers take a step to fill this gap, their focuses are on observable actions, i.e. with respect to either institutional ownership (Ilhan et al., 2023) or investors' public engagements, e.g. through submitting shareholder proposals (Baloria et al., 2019; Flammer et al., 2021) and public advocacy (Pawliczek et al., 2021). However, theoretical models and survey studies recognize that many of institutional investors' intervention on firms are through private interactions (Edmans and Manso, 2011; Solomon et al., 2011; McCahery et al., 2016, Edmans et al., 2019; Levit, 2019; Brav et al., 2022), yet little empirical evidence has been presented due to the unavailability of large-scale data of such activities. Utilizing the unique datasets of corporate visits in Chinese market, this paper fills the gap by showing that institutional investors make real efforts behind the scene to promote ESG disclosure through conducting costly corporate visits and discussing ESG issues with management.

Second, the paper contributes to the literature on firm information disclosure and in particular the growing body of literature on ESG disclosure. On one hand, the firm disclosure theory suggests that voluntary disclosure decisions depend on the benefits and costs associated with the disclosures (Bond and Goldstein, 2015; Goldstein and Yang, 2017; Abramova et al., 2020; Christensen et al., 2021). This paper adds to this stream of literature by showing that management responds (more actively) to institutional investors' (stronger) demands for ESG information and that the proprietary costs arising from disclosing ESG information reduce firms' tendency to disclose relevant information. The stream of literature on ESG disclosure, on the other hand, documents the benefits of better ESG disclosure in terms of increasing earnings forecasts accuracy, lowering stock crash risks, and reducing carbon emissions (e.g., Jouvenot and Krueger, 2020; Krueger et al., 2023) as well as the costs of ESG disclosure for corporates with respect to revealing proprietary information regarding firm fundamentals (Albuquerque et al., 2019; Pankratz and Schiller, 2020; Ilhan et al., 2023). This paper documents an additional benefit brought by greater ESG disclosure,

i.e. of reducing divergence in ESG ratings in the Chinese market, which is consistent with the conventional wisdom that greater disclosure reduces information asymmetry and consequently lower disagreements (e.g., Lang and Lundholm, 1996; Morgan, 2002; Hope, 2003).

Third, the paper adds to the understanding of the monitoring role of institutional investors. On one hand, the paper adds supportive evidence to the argument that institutional investors could perform effective governance through communications (Adams and Ferreira, 2007; McCahery et al., 2016; Levit, 2019) and that their threats of exit facilitate activism through communication (Firth et al., 2016; Levit, 2019; Cao et al., 2022). On the other hand, the paper adds to the literature specifically on on-site monitoring. Institutional investors' on-site monitoring is found to enhance plant-level investment and productivity (Giroud, 2013), spur corporate innovations (Bernstein et al., 2016; Jiang and Yuan, 2018), and increase corporate cash dividend payouts (Cao et al., 2022). The paper reveals another context where institutional investors' on-site monitoring could come into effect – to promote firm ESG disclosure. While the setting is in the Chinese market, I note that the observed positive effects of institutional investors on promoting ESG disclosure could be generalized to, or even become stronger in markets outside China, given that the role of institutional investors on monitoring managerial decisions in China has been perceived as less effective in comparison with its western counterparts (Jiang and Kim, 2015, 2020).

The rest of the paper proceeds as follows. Section 2 reviews relevant literature and develops hypotheses. Section 3 describes the data and variables. Section 4 documents the relationship between ESG-related corporate visits and ESG disclosure. Section 5 explores the mechanism through which ESG-related corporate visits affect ESG disclosure. Section 6 examines the impact of ESG disclosure on the divergence in ESG ratings. Section 7 concludes.

# 2. Literature review and hypothesis development

It is recognized that there is a considerable gap between investors' burgeoning demand for

ESG information and the supply of relevant information by firms (Amel-Zadeh and Serafeim, 2018; Gibson et al., 2020; Serafeim and Yoon, 2022; Agarwal et al., 2023). Given that public ESG information is the foundation of reliable ESG ratings which is the basis of sustainable investing (Chatterji et al., 2016; Berg et al., 2022; Christensen et al., 2022), to explore ways to promote ESG disclosure is a central concern for the sustainable finance community.

The existing literature suggests that the lack of ESG information can be largely attributed to its nature of non-financial information which lacks a standardized format of reporting as in the case of financial information (Elzahar et al., 2015; Baldini et al., 2018; Amel-Zadeh and Serefeim, 2018; Christensen et al., 2021). Although mandatory ESG disclosure regulations have been initiated in several countries (Chan et al, 2014; Talbot and Boiral, 2015; Krueger et al., 2023), the ESG information to be disclosed is still largely left to managers' discretion (Baldini et al., 2018; Tsang et al., 2022). In the presence of considerable costs associated with information disclosure (e.g. information processing and collection costs, proprietary and agency costs etc.), managers would be reluctant to adjust the contents and manners they disclose information (Chapman and Green, 2017; Abramavo et al., 2020), in particular when disclosing (additional) ESG information may set a precedent for future disclosure (Graham et al., 2005).

Against this backdrop, the pressure and information demand from market participants that have disciplining effects on manager decisions can be a powerful device to encourage firms to disclosure ESG information (Friedman et al., 2021). Consistently, Huang and Kung (2010) argue that the level of environmental disclosure is affected by demands from stakeholder groups such as governments, debtors, consumers, employees and shareholders. Likewise, Huang and Watson (2015) argue that external pressure from stakeholders affects CSR reporting.

Among these stakeholder groups, institutional investors are perceived as one particularly powerful market force to affect managerial decisions, given their active role in shareholder votes (Abramova et al., 2020) and possession of the option to sell shares due to dissatisfaction with firms which could lead to stock price decline (Admati and Pfleiderer., 2009; Edams and Manso, 2011; Dou et al., 2018). Consistently, numerous studies document that firms respond to the attention and demand from institutional investors, increasing the level of financial information disclosure (e.g., Bushee and Noe, 2000; Healy and Pelapu, 2001; Beyer et al., 2010; Boone and White, 2015; Bird and Karolyi, 2016; Dou et al., 2018; Abramova et al., 2020). In addition, institutional ownership improves firms' environmental and social performance (Dyck et al., 2019; Gantchev et al., 2022) and reduces facility toxic release (Kim et al., 2019).

In the context of ESG information disclosure, Ilhan et al. (2023) show that the level of institutional ownership is positively associated with firms' disclosure of climate risks. Flammer et al. (2021) document that larger number of environment-related proposals submitted by shareholders elicit greater disclosure of climate risk. Shareholder proposals could also motivate managers to disclose political spending information (Baloria et al., 2019). Pawliczek et al. (2021) find that BlackRock's portfolio firms disclose topics regarding environmental and regulatory factors similar to those discussed in its annual Dear CEO letter.

While these studies provide valuable insights into the role of institutional investors in promoting ESG disclosure, their focuses are on the observable actions. However, many interactions between institutional investors and firms during which investors perform monitoring occur behind the scenes. McCahery et al. (2016) stress the crucial role of private communications between investors and firms as a corporate governance mechanism, perhaps more important than previously thought. The effectiveness of the behind-the-scene soft activism by institutional investors (through, for instance, in-person discussions, and telephone calls etc.) on affecting managerial decisions are highlighted by extensive research (e.g., Brav et al., 2008; Becht et al., 2009; Dimson et al., 2015; Corum and Levit, 2019; Edmans et al., 2019; Levit, 2019; Zhang, 2023).

Private interactions with management can take many forms, with a crucial channel being corporate visits. Corporate visits, typically initiated by institutional investors, provide investors with the opportunities to observe firm operations, inspect production facilities, and discuss with managers (Cao et al., 2022). However, corporate visits activities are not required to be disclosed in most stock markets around the world, with the exception of the Shenzhen Stock Exchange (SZSE) in China. Therefore, the Chinese market provides a unique setting to perform large-scale empirical analysis of the effectiveness of corporate visits on affecting managerial decisions. Utilizing the dataset, Jiang and Yuan (2018) find that corporate visits lead to greater corporate dividend payout through the mechanism of disciplining management with exit threats.

In a similar vein, I conjecture that institutional investors are able to convey their burgeoning demand for ESG information through discussions with managers on ESG issues during corporate visits (i.e. ESG-related corporate visits), and that managers have incentives to respond to their needs in order to avoid the share selling by institutional investors which may lead to stock price decline. Thus, I propose the following hypothesis.

*H1*: Institutional investors' ESG-related corporate visits are associated with greater ESG disclosure.

I also conjecture that the above link should vary with ESG information demands from institutional investors and the level of proprietary costs faced by firms. First, institutional investors' stronger demand for ESG information should be associated with more intense monitoring on ESG disclosure. This is because without sufficient ESG disclosure, investors with stronger demand for ESG information would incur higher information acquisition costs which could erode their pecuniary benefits (Agarwal et al., 2023). Consistently, Ilhan et al. (2023) show that the groups of institutional investors that have a stronger demand for climate disclosure induce greater reporting

of climate risk. Second, besides external pressures and demands, firm disclosure theory suggests that firms' disclosure policy also depends upon the proprietary costs brought by the disclosure (Verrecchia, 1983; Healy and Palepu, 2001; Beyer et al., 2010; Lang and Sul, 2014; Bernard et al., 2018; Christensen et al., 2021). ESG information is documented to contain proprietary information about firm fundamentals (Dhaliwal et al., 2012; Albuquerque et al., 2019; Pankratz and Schiller, 2020; Ilhan et al., 2023) and the role of proprietary costs in impeding climate risk disclosure is confirmed in Ilhan et al. (2023). Following this line of thought, I propose the following hypothesis.

*H2*: The positive relationship between ESG-related corporate visits and firm ESG disclosure strengthens with institutional investors' information demands and weakens with firms' proprietary costs.

I also explore the mechanism through which ESG-related visits exert effects on firm ESG disclosure, and the outcomes to the sustainable finance market brought up by improvement in ESG disclosure. These will be discussed in detail in the empirical analysis sections.

# 3. Data and variable construction

#### 3.1 Data collection

Data for corporate visits during the period of 2013-2020 are collected from China Stock Market and Accounting Research (CSMAR). The Shenzhen Stock Exchange (SZSE) mandates that all listed firms disclose information for investor relation activities including corporate visits in their annual reports from 2009. CSMAR maintains complete records of the names of visiting institutions, dates, and meeting minutes of every visit since 2012.<sup>2</sup> As the paper focuses on the monitoring role of institutional investors, I keep the records only for the corporate visits conducted by fund companies, asset management companies, securities (brokers), insurance companies,

<sup>&</sup>lt;sup>2</sup> Table A.2 in the Appendix shows an example of the visit records, following Agarwal et al. (2023).

banks and trust companies, which in total conduct 83.5% of all the corporate visits on the market during the sample period.<sup>3</sup> To calculate the geographical distance between a visitor and the visited firm, for each of the 26,791 visiting institutions, I manually search the city it headquarters. I utilize the signatory to the United Nations Principle for Responsible Investment (UNPRI) as a proxy for an institutional investor's ESG awareness. To identify whether a visiting institution is a UNPRI-signatory, I manually match the names of the visiting institutions provided by CSMAR and those from the UNPRI website.<sup>4</sup>

Given that corporate visit data is available for SZSE-listed companies only, in the following analysis, the firm sample is restricted to SZSE-listed firms, excluding those listed on Shanghai Stock Exchange. For the sample of 2,343 SZSE-listed firms, I extract their daily stock returns, quarterly accounting data, the cities they headquarter, the industries they belong to, whether the firms have Big 4 auditors, and the yearly values for the proportion of shares held by institutional investors from CSMAR. Statistics show that out of the 2,343 firms, 2,062 are visited at least once in a calendar year by the sample institutional investors during the period of 2013-2020, translating to 48,265 corporate visit activities in total.

I use Bloomberg's annual ESG disclosure score as the proxy of firms' ESG disclosure, consistent with prior studies (e.g. Baldini et al., 2018; Christensen et al., 2022). According to Bloomberg, the disclosure scores range from 0-100% based on the amount of data points disclosed by the company and collected by Bloomberg from a variety of publicly available documents from sources, including company websites, exchanges and CDP responses.<sup>5</sup> I also collect the granular disclosure scores for the firms in each year for the three ESG pillars. Panel A of Table 1 reports the number of SZSE-listed firms in each year that have ESG disclosure score data during the period

<sup>&</sup>lt;sup>3</sup> My own calculations. Other institutions that conduct corporate visits include, for instance, rating agencies, law and accountancy firms, consulting firms, governments and universities etc.

<sup>&</sup>lt;sup>4</sup> See <u>https://www.unpri.org/signatories/signatory-resources/signatory-directory</u>

<sup>&</sup>lt;sup>5</sup> Also see a detailed description of the Bloomberg ESG disclosure score in Christensen et al. (2022).

of 2013-2020. Out of the initial sample of 2,343 firms, 698 firms have ESG disclosure scores, which constitutes the final sample of the main empirical analysis.

# 

#### **3.2 Variable construction and summary statistics**

3.2.1 ESG disclosure score, ESG-related corporate visits and firm-level characteristics

The dependent variable, a firm's ESG disclosure score (or E-/S-/G-pillar disclosure score) in a given year is denoted as *ESGD* (or *ED*, *SD* and *GD*, respectively). The independent variable of interest is a dummy *ESGvisit* indicating that a firm is visited and ESG issues are discussed during at least one visit in a calendar year. Similarly, the dummy *NESGvisit* indicates that a firm is visited and ESG issues are not discussed in any visits in a calendar year. To identify which visits are ESGrelated, I follow the method in Agarwal et al. (2023), defining an ESG-related visit if the visiting institutions pose at least one question containing ESG-issue keywords during a visit.<sup>6</sup> It is worth noting that to focus on the questions visitors pose rather than the answers the visited firms deliver ensures to capture the active decision of institutional investors to discuss ESG issues, reflecting their activism in monitoring the visited firms. Panel B of Table 1 reports the number of firms that host (non-) ESG-related visits by year. It shows that out of the 698 sample firms, 515 ever host ESG-related visits during the period of 2013-2020, with on average 35% of the sample firms hosting ESG-related visits in an individual year.

For each firm in each year, I construct control variables that are documented to affect firms' information disclosure (e.g., Francis and Wang, 2008; Huang and Kung, 2010; Abramova et al., 2020; Tsang et al., 2022) for which the definitions are presented in Table A.1 in the Appendix. Table 2 reports the descriptive statistics for these firm-level variables.

Panel A shows that the average (median) ESGD of the sample firms is 27.7 (27.2), with firms

<sup>&</sup>lt;sup>6</sup> Table A.3 in the Appendix shows the keywords used to identity E-/S-/G-issues, following Agarwal et al. (2023).

performing worst (best) in disclosing environmental (governance) information, indicated by the average score of 7.4 (62.9) for *ED* (*GD*).<sup>7</sup> On average, 34.6% of the firms host ESG-related visits and this number is slightly higher for non-ESG-related visits (36.4%). While *ESGD* does not differ statistically significantly between the firms that host ESG-related visits and host non-ESG-related visits (indicated by the statistical insignificance of the difference in means of 0.166), the *ESGD* in the next year (*ESGD\_lead*) is higher for the firms hosting ESG-related visits than for those hosting non-ESG-related visits (statistically significant at 1 percent). The outperformance in *ESGD\_lead* of firms that host ESG-related visits already provides supportive evidence for the hypothesis that firms' ESG disclosure improves after discussions of ESG issues with institutional investors during corporate visits.

## 

# 3.2.2 Proxy for information demand and proprietary cost

To test whether institutional investors' monitoring role in ESG disclosure varies with how much they demand ESG information, I construct four measures to proxy for their information demands. The first two measures capture the intensity of institutional investors proposing ESG-related questions to the visited firms. Specifically, the first is the total number of ESG-related visits (*#ESG-visits*) a firm hosts in a given year, and the second is the total number of ESG-related questions (*#ESG-qs*) a firm receives from all the corporate visits it hosts in a given year. The third measures the geographical distance between the visiting institutions and the visited firms. This measure could reflect institutional investors' information demands given the considerable costs incurred during distant on-site visits (Giroud, 2013; Bernstein et al., 2016; Cao et al., 2022). I first calculate the geographical distance between the cities where the visiting institution and the visited

<sup>&</sup>lt;sup>7</sup> The evidence is comparable to the statistics presented in Christensen et al. (2022) that the average (median) ESG disclosure score across firms in 69 countries during the period of 2004-2016 is 28.5 (25.6), and that firms worldwide, on average, tend to perform best in disclosing governance information and worst in disclosing environmental information.

firm headquarters for each ESG-related visit (for the visit with multiple visitors, the median geographical distance across all the visitors is calculated). Then, for a firm hosting multiple visits in a given year, I use the median geographical distance across all the ESG-related visits (*Geodis*) as a proxy for institutional investors' information demand. The fourth is the total number of UNPRI-signatory visitors a firm receives across all the ESG-related visits the firm hosts in a given year (*#UNPRI*). As signing the UNPRI is a public commitment to responsible investing, it is natural to expect that UNPRI-signatories would be more ESG aware and thus have a stronger intention to acquire ESG information from corporate visits in order to fulfil their commitment to responsible investing (Agarwal et al., 2023).

Numerous studies find that proprietary costs are negatively associated with disclosure levels (e.g., Verrecchia, 1983; Lang and Sul, 2014; Abramova et al., 2020; Ilhan et al., 2023) and in general relate industry concentration to greater proprietary costs and less disclosure (e.g., Harris, 1998; Bamber and Cheon, 1998; Botosan and Standord, 2005; Lang and Sul, 2014). To test the role of proprietary costs, I calculate the Herfindahl-Hirschman Index (*HHI*) measure based on firm annual revenues for the industry a given firm belongs to. Panel A of Table 3 reports the descriptive statistics for the variables defined above, to show the variation in information demands and in proprietary costs in relation to the firms that host ESG-related visits. It shows that the average (median) firm hosts 3.1 (2) ESG-related visits and the visitors on average pose 4.1 ESG-related questions to the firm is 912 (878) kilometers. For the majority of the firms hosting ESG-related visits, the visitors are not UNPRI-signatories (indicated by the median of zero for #*UNPRI*). Importantly, the statistics of the standard deviations and the values in the 10<sup>th</sup> and 90<sup>th</sup> percentiles indicate large variation across firms in all of the five measures.

Furthermore, Panel B shows that regardless of the information demand measures assessed, the

*ESGD* and *ESGD\_lead* for the high-demand group (from the high-tertile group) is larger than those for the low-demand group (from the low-tertile group), with the differences in means being statistically significant at 5 or 1 percent when *#ESG-visits*, *#ESG-qs*, and *#UNPRI* are assessed. The evidence suggests that institutional investors' stronger demand for ESG information is potentially associated with firms' greater ESG disclosure.

# 4. ESG-related corporate visits and firm ESG disclosure

# 4.1 Baseline results

To test the effect of ESG-related corporate visits on firms' ESG disclosure, I first employ pooled OLS regressions where the dependent variable is the ESG disclosure score in the next year (*ESGD\_lead*). The independent variable of interest is the dummy *ESGvisit* indicating a firm hosts at least one ESG-related visit in a given year, after controlling for various firm-level characteristics (i.e. *Size, ROA, Vol, Lev, Big4 and Insratio*), plus fixed effects of years, the industry a firm belongs to, and the city a firm headquarters.

Specifications (i) and (ii) in Panel A of Table 4 report the results when the city-fixed effects are excluded and included, respectively, in order to show whether the results hold after considering the location factors such as economic conditions and transportation convenience of the cities the firms headquarter. Regardless of the specification, the coefficient of *ESGvisit* is positive and statistically significant at 1 or 5 percent. Specifications (i) and (ii) suggest that hosting ESG-related visits is associated with a subsequent increase in the firm's ESG disclosure score of 0.816 or 0.591, respectively. In addition, both specifications show that the firms with larger market capitalization, larger institutional ownership and that have Big4 auditors tend to have higher future ESG disclosure scores, consistent with the findings in the existing literature (Huang and Kung, 2010; Baldini et al., 2018; Yu and Luu, 2021).

For comparison, I replace the dummy *ESGvisit* with *NESGvisit* to examine whether the corporate visits without discussions of ESG issues also have such monitoring role in promoting firms' ESG disclosures, with the results reported in Specifications (iii) and (iv) in Panel A. Both specifications show that the coefficients of *NESGvisit* are negative and statistically insignificant, suggesting no effect of non-ESG-related visits on firms' ESG disclosure. The findings still preserve when both *ESGvisit* and *NESGvisit* dummies are included in the regressions, as shown in Specifications (v) and (vi) of Panel A.

To test the robustness of the results, Panel B of Table 4 reports the analogous specifications to (i), (iii) and (v) in Panel A but adopting panel regressions with firm-fixed effects.<sup>8</sup> The results for *Size* and *Big4* preserve, while the statistical significance of the positive effect of institutional ownership weakens, and the positive effect of *ROA* on ESG disclosure scores becomes statistically significant. Importantly, the main finding of a statistically significantly positive effect of *ESGvisit* still preserves, regardless of whether controlling for the effect of *NESGvisit* in the specification or not.

Overall, the stark differences in the effects on firm ESG disclosure scores associated with the ESG-related and non-ESG-related visits highlight the importance of institutional investors discussing ESG issues during corporate visits in monitoring firms to promote their ESG disclosure practices.

#### 4.2 Addressing endogeneity concern

4.2.1 Nearest-neighbor matching and difference-in-differences approach

The baseline results are subject to the endogeneity concern that systematic differences among firms may drive both the propensity of a firm receiving ESG-related questions (i.e. hosting ESG-

<sup>&</sup>lt;sup>8</sup> The Hoechle method (Hoechle, 2007) with Driscoll–Kraay standard errors (Driscoll and Kraay, 1998) for unbalanced panels is adopted to adjust for heteroscedasticity, cross–sectional correlations and autocorrelation.

related corporate visits), and the firm's ESG disclosure. To mitigate the endogeneity issue, I employ the difference-in-differences (DiD) identification strategy after performing nearest-neighbor matching (NNM) on firm characteristics. To be specific, I first employ NNM with replacement and for each firm that ever host ESG-related visits during the sample period (i.e. the treatment group of firms), find one firm that never host ESG-related visits but share similar *ESGD* and *Size*, and operate in the same industry of the firm questioned in the year prior to the ESG-related visit (i.e. the control group of firms), in order to reduce the observable differences between the two groups of firms.<sup>9</sup> Then in order to rule out the effects of time-invariant unobservable differences between the treatment and the control groups, the matched sample is used for the following model of DiD estimation:

$$ESGD_{i,t+1} = \alpha + \beta Treat_{ESGi} \times Post_t + \gamma' X_{it} + \theta_i + \delta_t + \varepsilon_{i,t}$$
(1)

where  $ESGD_{i,t+1}$  represents firm *i*'s ESGD score in year t+1,  $Treat_{ESGi}$  indicates a firm coming from the treatment group (i.e. firms that ever hosts ESG-related visits),  $Post_t$  is the time indicator for the post-treatment years for both the treatment and control groups,  $X_{it}$  is a vector of firm-level characteristics in year *t* used in the baseline regressions,  $\theta_i$  and  $\delta_t$  represents firm- and yearfixed effects, respectively.

Specification (i) in Panel A of Table 5 reports the DiD estimation results. The positive effects of firm size, ROA and having Big 4 auditors on ESG disclosure observed in the baseline results still hold. Importantly, the coefficient of the DiD estimator, i.e. the interaction term  $Treat_{ESG} \times Post$  is positive and statistically significant at 5 percent, suggesting that the increase in the ESG disclosure score is larger for the treatment group than for the control group of firms by

<sup>&</sup>lt;sup>9</sup> When a firm hosts ESG-related visits in multiple years, its pre-ESG-visit years are defined as the years prior to the year when the earliest ESG-related visit occurred. Figure IB in Internet Appendix B shows the K density plots for the matching covariates (*ESGD* and *Size*) for the treated and control group of firms before and after adopting the NNM. It shows that the matching reduces the distributional difference in *ESGD* during the pre-treatment period between the two groups of firms, while the difference in *Size* remains at a minimal level after the matching.

0.27 following the ESG-related visits.

It is possible that institutional investors pose 'implicit' ESG-related questions during non-ESG-related corporate visits which may also have a monitoring effect on firms' ESG disclosure practices. Therefore, to rule out the possibility that the results are 'contaminated' by the effects of non-ESG-related visits, I exclude the firms that have hosted non-ESG-related visits before hosting ESG-related visits from the sample, and then repeat the NNM and DiD estimation. The results are presented in Specification (ii) in Panel A of Table 5. The positive and statistically significant coefficient of the interaction term confirms the finding observed in Specification (i). Moreover, the magnitude of its coefficient over doubles that in Specification (i), suggesting that the effect of ESG-related visits on firms' ESG disclosures becomes even stronger after removing the effect of non-ESG-related visits.

A key condition for validity of the DiD estimation is the parallel trend assumption. While the NNM on *ESGD* already reduces the pre-treatment difference in the outcome variable between the treatment and control groups of firms, to formally test the assumption and also to test the dynamics of the treatment effect, I estimate a variety of Eq. (1) by including dummies to trace out the year-by-year effects of ESG-related visits on firms' future ESG disclosure score, following the model used in Beck et al. (2010):

$$ESGD_{i,t+1} = \alpha + \beta_1 D_{it}^{-3} + \beta_2 D_{it}^{-2} + \dots + \beta_7 D_{it}^4 + \beta_8 D_{it}^5 + \gamma' X_{it} + \theta_i + \delta_t + \varepsilon_{i,t}$$
(2)

where the dummies ("D"s) equal zero except that  $D^{-j}$  equals one for firms in the *j*th year before the (earliest) ESG-related visits, and  $D^{+j}$  equals one for firms in the *j*th year after the (earliest) ESG-related visits. I exclude the year just prior to the ESG-related visits, thus estimating the dynamic effect of ESG-related visits on firms' future ESG disclosure scores relative to the year prior to the visits.

Panels A and B of Figure 1 plot the coefficients results and the 99% confidence intervals when

the firms that have hosted non-ESG-related visits before hosting ESG-related visits are included and excluded from the regressions, respectively. The two panels share similar patterns and highlight two key points: (i) the ESG disclosure scores do not differ statistically significantly between the treatment and the control groups of firms prior to ESG-related visits, and (ii) firms hosting ESG-related visits significantly increase their ESG disclosure in the following years. The results suggest that the parallel trend condition in absence of ESG-related visits is satisfied and that the effects of ESG-related visits could persist for years after the visits. One caveat to interpret the dynamic effects is that the persistent effect of the ESG-related visits in the following years may be driven by repeated ESG-related visits, if any, to a given firm, rather than the earliest one of question in the regression specifications.

For comparison, I repeat the analysis analogous to those in Panel A of Table 5, but examine the effects of non-ESG-related visits on firms' ESG disclosure. Specifically, the treatment group consists of the firms that ever host non-ESG-related visits (indicated by the dummy *Treat<sub>NESG</sub>*), and analogous NNM is performed to construct the control group of firms that never host non-ESGrelated visits. The matched sample is used for the DiD estimation. Panel B of Table 5 reports the results analogous to those in Panel A. It shows that no matter whether the effect of ESG-related visits is excluded (Specification (ii)) or not (Specification (i)), the coefficients of the interaction term *Treat<sub>NESG</sub>* × *Post* are statistically insignificant, indicating that non-ESG-related visits do not affect firms' ESG disclosure.

#### 

Overall, this section adopts the DiD identification strategy based on NNM and illustrates that firms increase their ESG disclosure after hosting ESG-related corporate visits more than the comparable firms that do not host ESG-related visits, reinforcing the main finding.

# 4.2.2 Heckman two-stage estimation

I employ a variety of the Heckman (1979) two-step estimation – the treatment effect model, as an alternative approach to mitigate the endogeneity issue associated with self-selection bias. In the first-step model, probit regression is adopted to estimate the probability a firm hosting ESG-related visits. In the second-step model, the inverse Mills ratio (IMR) obtained from first-step estimation is included as an additional independent variable in the specifications analogous to those in Table 4. To achieve first-step estimation, I construct an instrumental variable *#firms* measuring the number of firms being visited in the city a given firm headquarters in a given year.<sup>10</sup> This instrumental variable is chosen because location factors such as economic growth, transportation convenience and regional policy etc. are crucial for institutional investors to make corporate visit decisions (e.g., Giroud, 2013; Chen et al., 2022). Moreover, this variable is unlikely to affect ESG disclosure at firm level except via corporate visits.

Panels A and B of Table 6 report the results for the first-stage and the second-stage of the Heckman two-step regressions, respectively. Panel A shows that the firms that institutional investors have a higher propensity to visit and discuss ESG issues with are more likely to be located in cities with more corporate visit activities, indicated by the positive coefficient of Ln(#firms), statistically significant at 1 percent, making it a valid instrument variable for the Heckman (1979) adjustment procedure. Importantly, Panel B shows that the six coefficients of *ESGvisit* are all positive and statistically significant at 1 or 5 percent, after adding the *IMR* obtained from the first-stage estimation (Specifications (i)-(ii) and (iv)-(v) adopt pooled OLS regressions, and Specification (iii) and (vi) adopt panel regressions with firm-fixed effects), providing supportive evidence for the main finding.

<sup>&</sup>lt;sup>10</sup> I also use an alternative instrumental variable  $\#firms\_esg$  measuring the number of firms that host ESG-related corporate visits in the city a given firm headquarters in a given year. The results remain qualitatively similar and are reported in Table IB.1 in Internet Appendix B.

# 4.3 Effect by information demand and proprietary cost

To test the second hypothesis, I repeat the NNM and DiD estimation analogous to that in Specification (ii) in Panel A of Table 5 but using the subsamples created based on the various measures for information demands and proprietary costs defined in Section 3.2.2.<sup>11</sup> To start with, Table 7 reports the DiD estimation results for the subsamples of firms with ESG-related visits grouped by *#ESG-visits*, *#ESG-qs*, *Geodis* and *#UNPRI* in Panels A-D, respectively. In each panel, Specification (i) and (ii) report the results when the firms with the lowest and highest tertile of the grouping variable are assessed, respectively.

Panel A of Table 7 shows that the positive coefficient of the interaction term  $Treat_{ESG} \times Post$ is statistically significant at 1 percent when the firms with a large number of ESG-related visits are assessed (Specification (ii)) only. Moreover, the magnitude of the coefficient (1.086) doubles that (0.545) in Specification (i), suggesting that the increase in a visited firm's ESG disclosure after ESG-related visits is more prominent when the firm receives more ESG-related visits in a given year. Similar pattern is also observed when the firms are grouped by the number of ESG-related questions a firm receives during the corporate visits in a given year (#ESG-qs). Specifically, Panel B shows that the positive coefficient of the interaction term  $Treat_{ESG} \times Post$  is statistically significant at 5 percent when the firms receiving larger number of ESG-related questions are assessed (Specification (ii)) only, and that the magnitude of the coefficient (0.74) nearly doubles that (0.385) in Specification (i).

Given the considerable costs incurred during conducting geographically distant on-site visits (Giroud, 2013; Cheng et al., 2016; Cheng et al., 2019; Cao et al., 2022; Agarwal et al., 2023), the

<sup>&</sup>lt;sup>11</sup> I also run the DiD estimation analogous to that in specification (i) of Table 5 Panel A when the firms that have hosted non-ESG-related visits before ESG-related visits are not excluded from the sample. The results remain qualitatively similar and are reported in Table IB.2 in Internet Appendix B.

geographical distance could reflect institutional investors' information acquisition demands. Panel C of Table 7 reports the results when the firms with ESG-related visits are grouped by *Geodis*. It shows that the positive effect of ESG-related visits on firms' ESG disclosure is only observed for the firms that host corporate visits initiated by geographically distant institutional investors (Specification (ii)).

Moreover, Panel D illustrates the effect of demand for ESG information by examining visitors' ESG awareness. It shows that while both coefficients of the interaction term are positive and statistically significant at 10 percent, the coefficient magnitude in Specification (ii) is over four times that in Specification (i) (3.116 versus 0.66), suggesting that the effect of ESG-related visits on visited firms' ESG disclosure is stronger when more UNPRI-signatory institutional investors participate the visits.

# 

Another way to capture information demand from institutional investors is to examine their demand for the specific E-/S-/G-pillar information. Table 2 shows that firms' disclosure score is lowest (highest) for the E-pillar (G-pillar). Therefore, institutional investors are expected to have the strongest demand for the opaquest strand of information – environmental information. Therefore, I separately examine the effects of environment-related (social- or governance-related) corporate visits on firms' *ED* (*SD* or *GD*, correspondingly) in the next year. Specifically, I repeat the NNM and DiD estimation separately for the three ESG pillars. In the regressions, the indicator for the treatment group is replaced with *Treat<sub>E</sub>*, *Treat<sub>S</sub>* and *Treat<sub>G</sub>* indicating that a firm ever hosts E-/S-/G-related corporate visits, and the dependent variable is replaced with *ED\_lead*, *SD\_lead* and *GD\_lead*, respectively. The results are reported in Table 8. Panels A and B present the results when the firms that have hosted non-ESG-related visits before E-/S-/G-visits are included and excluded from the regressions, respectively.

The first column of Panel A shows a positive and statistically significant (at 5 percent) coefficient of *Treat<sub>E</sub>* x *Post*, suggesting that the increase in firms' environmental information disclosure is larger for the treatment group of firms than for the control group of firms by 1.11 following the E-related visits. However, such effects are not observed for the S- or G-related visits, as shown in the second and third column in Panel A. Moreover, the coefficient magnitude in Specification (i) is the largest among the three specifications and that for the G-related visits even becomes negative, albeit statistically insignificant. The results shown in Panel B after removing the effects of non-ESG-related visits remain qualitatively similar. The statistically insignificant results for the social-pillar disclosure is also consistent with the argument in Aluchna et al. (2022) that social issues, such as community involvement, human rights and employee relations are not major risk factors and therefore are not the prime concern for institutional investors. In summary, the evidence implies that the monitoring effect of institutional investors on firms' ESG information disclosure is most prominent for environmental information, which is the least transparent pillar.

To assess whether the observed effect varies across firms with different levels of proprietary costs, I repeat the NNM and DiD estimation for the firms operating in industries of low and high concentration separately. The results are reported in Table 9. Panels A and B present the results when the firms that have hosted non-ESG-related visits before ESG-related visits are included and excluded from the regressions, respectively. The two panels in Table 9 illustrate that the coefficients of *Treat*<sub>ESG</sub> x *Post* are positive and statistically significant at 1 or 10 percent for Specification (i) only, implying that institutional investors' monitoring role in pushing firms' ESG disclosure is only effective when the visited firms operate in industries of low proprietary costs.

# 

Overall, this section shows consistent evidence implying that institutional investors exert

stronger monitoring effects on firms' ESG disclosure when they have stronger demands for ESG information, whilst the effectiveness of their monitoring could be offset by firms' reluctance to disclose ESG information in the presence of high proprietary costs.

#### 5. Mechanism through threats of exit

To tighten the relationship between ESG-related visits and firm ESG disclosure further, in this section I explore the mechanism through which institutional investors' discussions of ESG issues with management exert effects on managerial decisions with respect to ESG disclosure.

Evidence is documented in the literature that trading is an effective corporate governance mechanism (Admati and Pfleiderer., 2009; Edams and Manso, 2011; Firth et al., 2016; Cvijanovic et al., 2022) and that soft activism can be facilitated if the activists have the option to exit (Levit, 2019; Cao et al., 2022). That is, institutional investors can *ex ante* discipline managerial behaviors by exit threats, assuming that the managers make efforts to avoid the share selling by institutional investors which may lead to stock price decline. More specifically, the survey conducted by McCahery et al. (2016) shows that 42% of the surveyed institutional investors believe that the exit threat is effective in disciplining managers and that 49% state that they had exited due to dissatisfaction with firm performance. Therefore, I conjecture that the exit threats could be an effective mechanism through which institutional investors' communications with firms during corporate visits exert real effects on firms' ESG disclosure.

While the *ex ante* threat of exit is unobservable, it is possible to test whether the exit threat is credible by observing the *ex post* trading behaviors of institutional investors after they conduct corporate visits, similar to Cao et al. (2022). To achieve this, I construct a subsample of institutional investors -- mutual funds for which portfolio holding data are disclosed regularly and publicly accessible.<sup>12</sup> Utilizing this dataset, I assess mutual funds' trading behaviors following ESG-related

<sup>&</sup>lt;sup>12</sup> Further details about the mutual fund data can be found in Internet Appendix A.

visits, using the specification similar to that in Cheng et al. (2019) and Cao et al. (2022) as follows:  $SHch_{fit} = \alpha + \beta ESGD_{it} + \gamma ESGvisit_{fit} + \theta ESGD_{it} \times ESGvisit_{fit} + \delta'X_{it} + \vartheta'F_{ft} + \tau_{fi} + \varphi_t + \varepsilon_{fit}$  (3) where  $SHch_{fit}$  is a fund f's holding change of a portfolio firm i, calculated as the number of shares of firm i the fund f holds at the end of year t minus the number of shares f holds at the end of year t-1. ESGD<sub>it</sub> is firm i's ESG disclosure score at the end of year t, ESGvisit<sub>fit</sub> is an indicator equal to one if firm i is visited by fund f in year t during which ESG issues are discussed.  $X_{it}(F_{ft})$  is a set of firm-level (fund-level) characteristics that could affect the share changes of mutual fund holdings, and the definition of the variables are presented in the Appendix.  $\tau_{fi}$  is fund-firm pair fixed effect, and  $\varphi_t$  is year fixed effect.

# 

To start with, Table 10 Specification (i) presents the results when the interaction term *ESGD* x *ESGvisit* is not included. The positive and statistically significant coefficient of *ESGvisit* suggests that mutual funds increase the holdings of shares of the firms after they discuss ESG issues during the corporate visits. It implies the existence of investment values of the information acquired from ESG-related corporate visits, consistent with the finding in Agarwal et al. (2023). It also adds to the evidence that corporate visits in general could create information advantages and affect mutual funds' trading behaviors (Cheng et al., 2016; Liu et al., 2017; Han et al., 2018). Moreover, the coefficient of *ESGD* (0.011) is positive and statistically significant at 1 percent, suggesting the preference of mutual funds to firms with better ESG disclosures.

Specification (ii) in Table 10 further includes the interaction terms of *ESGvisit* with *ESGD* in the regression, and shows a positive and statistically significant coefficient (at 5 percent) for *ESGvisit* x *ESGD*. It suggests that visiting funds sell more shares (or purchase less) if they find firms performing badly in ESG disclosures, relative to the funds that did not conduct ESG-related visits prior to trading.

For comparison, I repeat the analyses analogous to Specifications (i) and (ii) in Table 10 but replace the *ESGvisit* indicator with the *NESGvisit* indicator equal to one if a firm is visited by a fund in a given year during which only non-ESG-issues are discussed and zero otherwise.<sup>13</sup> Specifications (iii) and (iv) report the results, showing that while conducting non-ESG-related corporate visits also generate private information for mutual funds and increase funds' purchases of firm shares following the visits (indicated by the positive coefficient of *NESGvisit*, statistically significant at 1 percent in Specification (iii)), non-ESG-related visits do not strengthen the positive relationship between *ESGD* and *SHch* (indicated by the statistically insignificant coefficient of *NESGvisit* x *ESGD*).

Overall, the evidence in this section confirms the creditability of mutual funds' exit threats, in support of the conjecture that the threat of exit could be one mechanism through which institutional investors' discussions of ESG issues during corporate visits promote firms' ESG disclosures.

# 6. Consequences of greater ESG disclosure

The previous sections establish that institutional investors' monitoring activities through private communications with firms promote ESG disclosure. Then the question follows of what consequences the enhancement of ESG disclosure could bring up to the broad market participants. Conventional wisdom suggests that greater disclosure reduces disagreement (e.g., Lang and Lundholm 1996; Hope 2003), and in the ESG information context, it is argued that sufficient public disclosure of ESG information is the foundation for rating agencies to assign firms reliable ESG scores and to reduce the divergence in ESG ratings across rating providers (Chatterji et al., 2016; Berg et al., 2022). Following this line of thought, this section examines whether the divergence across ESG ratings varies with ESG disclosure.

To quantify ESG rating divergence, I retrieve firms' ESG score data in each year from four

<sup>&</sup>lt;sup>13</sup> For Specifications (i) and (ii), the firms having non-ESG-related corporate visits are excluded from the regressions. Similarly, for Specifications (iii) and (iv), the firms having ESG-related corporate visits are excluded from the regressions.

main ESG rating providers for Chinese listed firms -- Huazheng, Syntao Green Finance, Bloomberg and S&P.<sup>14</sup> Panel A of Table IB.3 in Internet Appendix presents the pairwise Pearson correlations between the four ESG performance measures. It shows that the correlations of the rating pairs range from -0.013 (for S&P – Huazheng pair) to 0.412 (for Bloomberg – Syntao pair), confirming the low consistency in the ESG ratings across different rating providers.

To measure ESG rating divergence, I first calculate firms' percentile ranking according to each of the four ESG ratings, and then for each pair of ratings in each year (e.g., Huazheng - Syntao, Huazheng - Bloomberg, Bloomberg – SP) calculate the square of the difference in percentile ranking based on the two ratings. The conversion from ESG scores to percentile rankings is done to remove the effect of scaling and make the divergences across different rating pairs comparable. The calculation gives me six measures for divergence for the six pairs of ratings, denoted as *HZ\_syn*, *HZ\_Blbg*, *HZ\_SP*, *Blbg\_syn*, *SP\_syn* and *SP\_Blbg*. Panel B of Table IB.3 in Internet Appendix presents the descriptive statistics for the six divergence measures.

To examine the effect of ESG disclosure on ESG rating divergence, I run regressions of firms' ESG rating divergence measures on *ESGD*, after controlling for firm-level characteristics same as those used in Table 4, plus year- and firm-fixed effects. Given that it may take time for rating agencies to collect public ESG information to adjust their ratings, firms' ESG rating divergence in the next year and in two years is used as the dependent variable, with the results reported in Panels A and B of Table 11, respectively.

#### 

The six columns of Table 11 present the results when the six rating divergence measures are

<sup>&</sup>lt;sup>14</sup> Huazheng and Syntao rating has gained popularity in studies on ESG performance in China (e.g., Broadstock et al., 2021; Zhang al., 2023). See rating methodology details https://www.chindices.com/files/Sinoet at Securities%20Index%20ESG%20Ratings%20Methodology.pdf (downloaded Aug Also in 2023). see http://syntaogf.com/Menu EN.asp?ID=34 for detailed information of the rating methodology of SynTao. Further details about the four ESG scores can be found in Internet Appendix A.

the dependent variables, respectively. The coefficient of *ESGD* is negative in 10 out of the 12 specifications, with 6 of them being statistically significant at 1 or 10 percent, providing consistent evidence that greater ESG disclosure is associated with lower ESG rating divergence. It is worth noting that Christensen et al. (2022) finds a positive relationship between ESG disclosure and ESG rating divergence in a cross-country analysis, arguing that the subjective nature of ESG information makes ESG disclosure expand opportunities for different interpretations of information. While Table 11 shows finding for the Chinese market opposite to Christensen et al. (2022), the results are in line with the conventional wisdom that greater disclosure reduces disagreement (e.g., Lang and Lundholm, 1996; Hope, 2003). The discrepancy in results calls for caution when examining the ESG disclosure-ESG rating relationship in different markets.

# 7. Conclusion

Utilizing the unique dataset of institutional investors' corporate visit activities in the Chinese market covering the period of 2013-2020, I find consistent and strong evidence that institutional investors' discussions of ESG-related issues during the corporate visits lead to greater ESG disclosure of the visited firms. To the best of my knowledge, this is the first study providing direct large-scale evidence that institutional investors' behind-the-scene monitoring activities elicit greater ESG disclosure.

Consistent with firm disclosure theory, this monitoring effect strengthens when institutional investors have stronger demands for ESG information and weakens when corporates operate in the industries with high proprietary costs. In addition, I document evidence that institutional investors' threat of exit is a potential mechanism through which their soft activism via private communications exerts effects on managerial decisions of ESG disclosure. Lastly, to understand the consequences of the effective monitoring of institutional investors on promoting ESG disclosure, I document that greater ESG disclosure is linked with a reduction in the divergence of

ESG ratings by different providers.

The paper reveals that institutional investors make real efforts behind the scene, e.g. conducting costly corporate visits and communicating with corporates regarding ESG issues, to promote corporates' ESG disclosure practices. The finding highlights that in addition to the various regulations and initiatives at the global and national level to promote mandatory ESG disclosure, the pressures and demands from market participants, particularly institutional investors, could be an effective stimulator of ESG disclosure. Moreover, the improvement in corporate ESG disclosure potentially brings benefits to the sustainable investing community at large, i.e. reduction in divergence in ESG ratings. Collectively, the paper highlights the crucial role of institutional investors in eliciting corporate ESG disclosure and pushing the ESG agenda.

# Reference

Abramova, I., Core, J. E., Sutherland, A., 2020. Institutional investor attention and firm disclosure. The Accounting Review, 95(6), 1-21.

Adams, R. B., Ferreira, D., 2007. A theory of friendly boards. The Journal of Finance, 62(1), 217-250.

Admati, A. R., Pfleiderer, P., 2009. The "Wall Street Walk" and shareholder activism: Exit as a form of voice. The Review of Financial Studies, 22(7), 2645-2685.

Agarwal, S., Sulaeman, J., Zhang, Y. and Zong, Z., 2023. Does the Commitment to Responsible Investing Alter Investment Activities? Available at SSRN 4568714.

Akins, B., 2018. Financial reporting quality and uncertainty about credit risk among ratings agencies. The Accounting Review, 93(4), 1-22.

Albuquerque, R., Koskinen, Y., Zhang, C., 2019. Corporate social responsibility and firm risk: Theory and empirical evidence. Management Science, 65(10), 4451-4469. Aluchna, M., Roszkowska-Menkes, M., Kamiński, B., Bosek-Rak, D., 2022. Do institutional investors encourage firm to social disclosure? The stakeholder salience perspective. Journal of Business Research, 142, 674-682.

Amel-Zadeh, A., Serafeim, G., 2018. Why and how investors use ESG information: Evidence from a global survey. Financial analysts journal, 74(3), 87-103.

Avramov, D., Cheng, S., Lioui, A., Tarelli, A., 2022. Sustainable investing with ESG rating uncertainty. Journal of Financial Economics, 145(2), 642-664.

Baldini, M., Maso, L. D., Liberatore, G., Mazzi, F., Terzani, S., 2018. Role of country-and firm-level determinants in environmental, social, and governance disclosure. Journal of Business Ethics, *150*, 79-98.

Baloria, V. P., Klassen, K. J., Wiedman, C. I., 2019. Shareholder activism and voluntary disclosure initiation: The case of political spending. Contemporary Accounting Research, 36(2), 904-933.

Bamber, L. S., Cheon, Y. S., 1998. Discretionary management earnings forecast disclosures: Antecedents and outcomes associated with forecast venue and forecast specificity choices. Journal of Accounting Research, 36(2), 167-190.

Becht, M., Franks, J., Mayer, C., Rossi, S., 2009. Returns to shareholder activism: Evidence from a clinical study of the Hermes UK Focus Fund. The Review of Financial Studies, 22(8), 3093-3129.

Beck, T., Levine, R., Levkov, A., 2010. Big bad banks? The winners and losers from bank deregulation in the United States. The Journal of Finance, 65(5), 1637-1667.

Berg, F., Koelbel, J. F., Rigobon, R., 2022. Aggregate confusion: The divergence of ESG ratings. Review of Finance, 26(6), 1315-1344.

Bernard, D., Burgstahler, D., Kaya, D., 2018. Size management by European private firms to

minimize proprietary costs of disclosure. Journal of Accounting and Economics, 66(1), 94-122.

Bernstein, S., Giroud, X., Townsend, R. R. 2016. The impact of venture capital monitoring. Journal of Finance, 71(4), 1591-1622.

Beyer, A., Cohen, D. A., Lys, T. Z., Walther, B. R., 2010. The financial reporting environment: Review of the recent literature. Journal of Accounting and Economics, 50(2-3), 296-343.

Bird, A., Karolyi, S. A., 2016. Do institutional investors demand public disclosure? The Review of Financial Studies, 29(12), 3245-3277.

Bond, P., Goldstein, I., 2015. Government intervention and information aggregation by prices. The Journal of Finance, 70(6), 2777-2812.

Bonsall, S. B., Miller, B. P., 2017. The impact of narrative disclosure readability on bond ratings and the cost of debt. Review of Accounting Studies, 22, 608-643.

Boone, A. L., White, J. T., 2015. The effect of institutional ownership on firm transparency and information production. Journal of Financial Economics, 117(3), 508-533.

Botosan, C. A., Stanford, M., 2005. Managers' motives to withhold segment disclosures and the effect of SFAS No. 131 on analysts' information environment. The Accounting Review, 80(3), 751-772.

Brav, A., Dasgupta, A., Mathews, R., 2022. Wolf pack activism. Management Science, 68(8), 5557-5568.

Broadstock, D. C., Chan, K., Cheng, L. T., Wang, X. (2021). The role of ESG performance during times of financial crisis: Evidence from COVID-19 in China. Finance research letters, 38, 101716.

Bushee, B. J., Noe, C. F., 2000. Corporate disclosure practices, institutional investors, and stock return volatility. Journal of Accounting Research, 171-202.

Cao, J., Wang, H., Zhou, S., 2022. Soft activism and corporate dividend policy: Evidence from

institutional investors site visits. Journal of Corporate Finance, 75, 102221.

Chan, M. C., Watson, J., Woodliff, D., 2014. Corporate governance quality and CSR disclosures. Journal of Business Ethics, 125, 59-73.

Chapman, K., Green, J. R., 2018. Analysts' influence on managers' guidance. The Accounting Review, 93(1), 45-69.

Chatterji, A. K., Durand, R., Levine, D. I., Touboul, S.,2016. Do ratings of firms converge? Implications for managers, investors and strategy researchers. Strategic Management Journal, 37(8), 1597-1614.

Chen, H., Qu, Y., Shen, T., Wang, Q., Xu, D. X., 2022. The geography of information acquisition. Journal of Financial and Quantitative Analysis, 57(6), 2251-2285.

Cheng, Q., Du, F., Wang, X., Wang, Y., 2016. Seeing is believing: Analysts' corporate site visits. Review of Accounting Studies 21 (4), 1245-1286.

Cheng, Q., Du, F., Wang, B. Y., Wang, X., 2019. Do corporate site visits impact stock prices?. Contemporary Accounting Research 36(1), 359-388.

Christensen, H. B., Hail, L., Leuz, C., 2021. Mandatory CSR and sustainability reporting: Economic analysis and literature review. Review of Accounting Studies, 26(3), 1176-1248.

Christensen, D. M., Serafeim, G., Sikochi, A., 2022. Why is corporate virtue in the eye of the beholder? The case of ESG ratings. The Accounting Review, 97(1), 147-175.

Corum, A. A., Levit, D., 2019. Corporate control activism. Journal of Financial Economics, 133(1), 1-17.

Cvijanović, D., Dasgupta, A., Zachariadis, K. E., 2022. The Wall Street stampede: Exit as governance with interacting blockholders. Journal of Financial Economics, 144(2), 433-455.

Dhaliwal, D. S., Radhakrishnan, S., Tsang, A., Yang, Y. G., 2012. Nonfinancial disclosure and analyst forecast accuracy: International evidence on corporate social responsibility disclosure. The

Accounting Review, 87(3), 723-759.

Dimson, E., Karakaş, O., Li, X., 2015. Active ownership. The Review of Financial Studies, 28(12), 3225-3268.

Driscoll, J.C., Kraay, A.C., 1998. Consistent covariance matrix estimation with spatially dependent panel data. Review of Economics and Statistics 80, 549–560.

Dou, Y., Hope, O. K., Thomas, W. B., Zou, Y., 2018. Blockholder exit threats and financial reporting quality. Contemporary Accounting Research, 35(2), 1004-1028.

Dyck, A., Lins, K. V., Roth, L., Wagner, H. F., 2019. Do institutional investors drive corporate social responsibility? International evidence. Journal of Financial Economics, 131(3), 693-714.

Edmans, A., Levit, D., Reilly, D., 2019. Governance under common ownership. The Review of Financial Studies, 32(7), 2673-2719.

Edmans, A., Manso, G., 2011. Governance through trading and intervention: A theory of multiple blockholders. The Review of Financial Studies, 24(7), 2395-2428.

Elzahar, H., Hussainey, K., Mazzi, F., Tsalavoutas, I., 2015. Economic consequences of key performance indicators' disclosure quality. International Review of Financial Analysis, 39, 96-112.

EY., 2018. Does your nonfinancial reporting tell your value creation story? Industry Study. EY Climate Change and Sustainability Services.

Firth, M., Gao, J., Shen, J., Zhang, Y., 2016. Institutional stock ownership and firms' cash dividend policies: Evidence from China. Journal of Banking & Finance, 65, 91-107.

Flammer, C., Toffel, M. W., Viswanathan, K., 2021. Shareholder activism and firms' voluntary disclosure of climate change risks. Strategic Management Journal, 42(10), 1850-1879.

Francis, J. R., Wang, D., 2008. The joint effect of investor protection and Big 4 audits on earnings quality around the world. Contemporary Accounting Research, 25(1), 157-191.

Friedman, H. L., Heinle, M. S., Luneva, I. M., 2021. A theoretical framework for ESG

reporting to investors. Available at SSRN 3932689.

Gantchev, N., Giannetti, M., Li, R., 2022. Does money talk? Divestitures and corporate environmental and social policies. Review of Finance, 26(6), 1469-1508.

Gibson, R., Krueger, P., Mitali, S. F., 2020. The sustainability footprint of institutional investors: ESG driven price pressure and performance. Swiss Finance Institute Research Paper, 17-05.

Giroud, X. 2013. Proximity and investment: Evidence from plant-level data. Quarterly Journal of Economics, 128(2), 861-915.

Goldstein, I., Yang, L., 2017. Information disclosure in financial markets. Annual Review of Financial Economics, 9, 101-125.

Graham, J. R., Harvey, C. R., Rajgopal, S., 2005. The economic implications of corporate financial reporting. Journal of Accounting and Economics, 40(1-3), 3-73.

Han, B., Kong, D., Liu, S., 2018. Do analysts gain an informational advantage by visiting listed companies?. Contemporary Accounting Research 35(4), 1843-1867.

Harris, M. S., 1998. The association between competition and managers' business segment reporting decisions. Journal of Accounting Research, 36(1), 111-128.

Healy, P. M., Palepu, K. G., 2001. Information asymmetry, corporate disclosure, and the capital markets: A review of the empirical disclosure literature. Journal of Accounting and Economics, 31(1-3), 405-440.

Heckman, J. J., 1979. Sample selection bias as a specification error. Econometrica: Journal of the Econometric Society, 153-161.

Hoechle, D., 2007. Robust standard errors for panel regressions with cross-sectional dependence. Stata Journal 7(3), 281–312.

Hope, O. K., 2003. Accounting policy disclosures and analysts' forecasts. Contemporary

Accounting Research, 20(2), 295-321.

Huang, C. L., Kung, F. H., 2010. Drivers of environmental disclosure and stakeholder expectation: Evidence from Taiwan. Journal of Business Ethics, 96, 435-451.

Huang, X. B., Watson, L., 2015. Corporate social responsibility research in accounting. Journal of Accounting literature, 34(1), 1-16.

Ilhan, E., Krueger, P., Sautner, Z., Starks, L. T., 2023. Climate risk disclosure and institutional investors. The Review of Financial Studies, 36(7), 2617-2650.

Jiang, F., Kim, K. A., 2015. Corporate governance in China: A modern perspective. Journal of Corporate Finance, 32, 190-216.

Jiang, F., Kim, K. A. 2020. Corporate governance in China: A survey. Review of Finance, 24(4), 733-772.

Jiang, X., Yuan, Q., 2018. Institutional investors' corporate site visits and corporate innovation. Journal of Corporate Finance, 48, 148-168.

Jouvenot, V., Krueger, P., 2020. Reduction in corporate greenhouse gas emissions under prescriptive disclosure requirements. Unpublished working paper. University of Geneva.

Kim, I., Wan, H., Wang, B., Yang, T., 2019. Institutional investors and corporate environmental, social, and governance policies: Evidence from toxics release data. Management Science, 65(10), 4901-4926.

Krueger, P., Sautner, Z., Tang, D. Y., Zhong, R., 2023. The effects of mandatory ESG disclosure around the world. European Corporate Governance Institute–Finance Working Paper, (754), 21-44.

Lang, M. H., Lundholm, R. J., 1996. Corporate disclosure policy and analyst behavior. The Accounting Review, 467-492.

Lang, M., Sul, E., 2014. Linking industry concentration to proprietary costs and disclosure:

Challenges and opportunities. Journal of Accounting and Economics, 58(2-3), 265-274.

Levit, D., 2019. Soft shareholder activism. The Review of Financial Studies, 32(7), 2775-2808.

Liu, S., Dai, Y., Kong, D., 2017. Does it pay to communicate with firms? Evidence from firm site visits of mutual funds. Journal of Business Finance and Accounting 44(5-6), 611-645.

McCahery, J. A., Sautner, Z., Starks, L. T., 2016. Behind the scenes: The corporate governance preferences of institutional investors. The Journal of Finance, 71(6), 2905-2932.

Morgan, D. P., 2002. Rating banks: Risk and uncertainty in an opaque industry. American Economic Review, 92(4), 874-888.

Pankratz, M., and Christoph M.S., 2020, Climate change and adaptation in global supplychain networks, Working Paper, University of California at Los Angeles.

Pawliczek, A., Skinner, A. N., Wellman, L. A., 2021. A new take on voice: the influence of BlackRock's 'Dear CEO'letters. Review of Accounting Studies, 26(3), 1088-1136.

PRI, 2023: PRI quarterly signatory update. Available at: https://www.unpri.org/signatories/signatory-resources/quarterly-signatory-update

Serafeim, G., Yoon, A., 2022. Stock price reactions to ESG news: The role of ESG ratings and disagreement. Review of Accounting Studies, 1-31.

Solomon, J. F., Solomon, A., Norton, S. D., Joseph, N. L., 2011. Private climate change reporting: an emerging discourse of risk and opportunity? Accounting, Auditing & Accountability Journal, 24(8), 1119-1148.

Talbot, D., Boiral, O., 2015. Strategies for climate change and impression management: A case study among Canada's large industrial emitters. Journal of Business Ethics, 132, 329-346.

Tsang, A., Frost, T., Cao, H., 2022. Environmental, social, and governance (ESG) disclosure: A literature review. The British Accounting Review, 101149.

Verrecchia, R. E., 1983. Discretionary disclosure. Journal of Accounting and Economics, 5,

179-194.

Yu, E. P. Y., Van Luu, B., 2021. International variations in ESG disclosure–do cross-listed companies care more? International Review of Financial Analysis, 75, 101731.

Zhang, R. X., 2023. Do Managers learn from institutional investors through direct interactions? Journal of Accounting and Economics, 75(2-3), 101554.

Zhang, N., Zhang, Y., Zong, Z., 2023. Fund ESG performance and downside risk: Evidence from China. International Review of Financial Analysis 86, 102526.

Figure 1. Tests for parallel trend assumption and dynamic effects of ESG-related visits on ESG disclosure.



Panel A. Test corresponding to the Specification (i) in Panel A of Table 5

Panel B. Test corresponding to the Specification (ii) in Panel A of Table 5



Panel A. Number of SZSE-listed firms and firms with valid ESG disclosure score data								
Year	#firms	#firms with ESGD score	#firms with ED score	#firms with SD score	#firms with GD score			
2013	1.405	412	396	409	412			
2014	1,542	429	412	427	429			
2015	1,716	581	565	580	581			
2016	1,825	579	563	578	579			
2017	2,029	494	494	494	494			
2018	2,072	520	520	520	520			
2019	2,094	573	573	573	573			
2020	2,151	573	573	573	573			
Total	2,343	698	683	696	698			

Table 1. Sample construction.

Panel B. Number of sample firms that have ESG-related visits and non-ESG-related visits #firms hosting

Year	# Sample firms	#firms hosting ESG-related visits	non-ESG-related visits only	
201	3 412	161	154	
201	4 429	187	137	
201	5 581	211	222	
201	6 579	211	236	
201	7 494	187	184	
201	8 520	179	196	
201	9 573	164	200	
202	0 573	138	185	
Total	1 698	515	546	

	Panel A. All firms				Panel B. Firms hosting ESG-related visits				Panel C. Fin non-ESG-re	rms hosting elated visits		T-test on the dif means	ferences in	
	Mean	Median	Std Dev	Obs.	Mean	Median	Std Dev	Obs.	Mean	Median	Std Dev	Obs.	ESG-NESG	p-value
ESGvisit	0.346	0.000	0.476	4,161										
NESGvisit	0.364	0.000	0.481	4,161										
ESGD	27.673	27.160	7.617	4,161	27.723	27.231	8.105	1,438	27.557	27.110	7.432	1,514	0.166	0.562
ESGD_lead	28.486	27.854	7.380	3,692	29.018	28.186	7.845	1,356	28.163	27.548	7.131	1,364	0.856***	0.003
ED	7.389	1.752	10.673	4,096	8.579	2.416	11.569	1,412	7.055	1.268	10.596	1,493	0.979**	0.015
ED_lead	7.674	1.752	10.956	3,645	8.988	2.416	11.785	1,338	7.192	1.571	10.599	1,346	1.387***	0.001
SD	13.907	12.281	6.914	4,154	14.563	12.817	6.980	1,437	13.638	12.001	6.944	1,512	0.925***	0.000
SD_lead	14.049	12.394	6.926	3,688	14.855	13.029	7.163	1,355	13.718	12.001	6.781	1,364	1.137***	0.000
GD	62.924	64.419	12.737	4,161	62.134	64.419	13.847	1,438	63.179	64.419	12.517	1,514	-1.045**	0.031
GD_lead	64.704	66.556	11.633	3,692	64.867	67.550	12.375	1,356	64.508	66.346	11.506	1,364	0.359	0.433
Size (bil														
CNY)	17.203	9.657	36.179	4,161	17.365	10.360	38.386	1,438	18.104	10.199	32.754	1,514	-0.739	0.573
ROA	0.031	0.026	0.041	4,161	0.038	0.033	0.035	1,438	0.034	0.028	0.041	1,514	0.005***	0.001
Vol	0.027	0.024	0.014	4,160	0.027	0.024	0.016	1,438	0.027	0.025	0.012	1,514	0.000	0.528
Lev	1.218	0.789	2.812	4,161	1.016	0.742	0.946	1,438	1.132	0.743	3.636	1,514	-0.116	0.241
Big4	0.072	0.000	0.259	4,161	0.079	0.000	0.270	1,438	0.061	0.000	0.239	1,514	0.019**	0.049
Instatio	48.353	49.679	23.615	4,159	48.458	50.205	24.536	1,437	47.632	48.368	23.720	1,514	0.825	0.353
Firms.		69	98			51	5			54	6			

Table 2. Descriptive statistics of firm characteristics.

*Notes*: Panel A reports the statistics calculated across all the sample firms, Panel B reports the statistics calculated across the firms that host at least one ESG-related visit in a given year, and Panel C reports the statistics calculated across the firms that host only non-ESG-related visits in a given year. ESGvisit is a dummy indicating that a firm is visited and ESG issues are discussed during at least one visit in a calendar year. NESGvisit is a dummy indicating that a firm is visited and ESG issues are not discussed in any visits in a calendar year. ESGD is a firm's ESG disclosure score in a given year, and ED (SD or GD) is the E- (S- or G-) pillar disclosure score in a given year. ESGD\_lead, ED\_lead, SD\_lead and GD\_lead is the value of ESDG, ED, SD, and GD, correspondingly in the next year. Size is the market capitalization of a firm at the end of a calendar year. Vol is a firm's average quarterly volatility of stock returns in a calendar year where quarterly volatility is calculated as the standard deviation of daily returns across a calendar quarter. ROA is a firms' annual return on assets calculated as the average ROA across the quarters in a calendar year. Lev is a firm's leverage ratio calculated as the ratio of debt to equity at the end of a calendar year. Big4 is a dummy indicating that a firm has Big 4 auditors in a given year. Instatio is the percentage of institutional ownership in the firm at the end of a year. \*\*\*, \*\*, \* are 1, 5 and 10 percent statistical significance respectively.

Table 3. Descriptive statistics of the proxy for information demand and proprietary costs for the firms hosting ESG-related visits (Panel A), and of *ESGD* and *ESGD\_lead* for the firms by groups of information demands and proprietary costs (Panel B).

Panel A. Proxy for information demand and disclosure costs									
	Mean	Median	Std Dev	p10	p90	Obs.			
#ESG-visits	3.054	2.000	3.876	1.000	7.000	1,438			
#ESG-qs	4.074	2.000	6.982	1.000	8.000	1,438			
Geodis (km)	911.951	878.258	519.133	285.371	1,546.088	1,437			
#UNPRI	0.879	0.000	6.299	0.000	2.000	1,438			
HHI	0.045	0.021	0.066	0.004	0.117	1,438			

Panel B. ESGD and ESGD\_lead for firms by groups of information demand and proprietary costs

	L	Low tertile of grouping variable				High tertile of grouping variable				T-test on difference in means	
	Mean	Median	Std Dev	Obs.	Mean	Median	Std Dev	Obs.	Tert1-Tert3	p-value	
ESGD											
by #ESG-visits	27.483	27.130	7.657	641	28.549	27.864	8.657	509	-1.066**	0.027	
by #ESG-qs	27.361	27.080	7.552	567	28.723	28.176	8.702	440	-1.362***	0.008	
by Geodis	27.504	27.382	7.562	429	27.829	27.216	8.280	524	-0.325	0.531	
by #UNPRI	26.267	26.019	7.659	1,122	32.891	30.646	7.524	316	-6.624***	0.000	
by HHI	27.725	27.140	7.713	502	27.812	27.321	8.555	473	-0.087	0.867	
ESGD_lead											
by #ESG-visits	28.482	27.995	7.140	584	29.812	28.599	8.331	501	-1.330***	0.005	
by #ESG-qs	28.319	27.955	7.096	519	30.131	28.785	8.397	432	-1.811***	0.000	
by Geodis	28.591	28.186	7.190	409	29.172	27.774	8.222	477	-0.581	0.267	
by #UNPRI	27.800	27.402	7.365	1,099	34.230	32.421	7.710	257	-6.431***	0.000	
by HHI	29.209	28.176	7.479	468	28.818	27.784	8.420	441	0.390	0.460	

Notes: #ESG-visits is the total number of ESG-related visits a firm hosts in a given year, #ESG-qs is the total number of ESG-related questions a firm receives from all the ESG-related visits it hosts in a given year. Geodis is the geographical distance between the visiting institutions and the visited firms. #UNPRI is the total number of UNPRI-signatory visitors a firm receives across all the ESG-related visits the firm hosts in a given year. HHI is the Herfindahl-Hirschman Index calculated based on firm annual revenues for the industry a given firm belongs to. ESGD is a firm's ESG disclosure score in a given year. ESGD lead is a firms' ESGD in the next year.

			Panel A. P	ooled OLS			Pane	el B. Firm fixed e	ffects
	(i)	(ii)	(iii)	(iv)	(v)	(vi)	(i)	(ii)	(iii)
ESGvisit	0.816***	0.591**			1.078***	0.829**	0.257***		0.412***
	(0.003)	(0.016)			(0.003)	(0.017)	(0.000)		(0.001)
NESGvisit			-0.216	-0.148	0.426	0.365		-0.052	0.214*
			(0.358)	(0.495)	(0.187)	(0.245)		(0.371)	(0.051)
Size	53.260***	50.509***	53.047***	50.285***	53.035***	50.311***	34.628***	34.621***	34.405***
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
ROA	3.547	2.565	4.777	3.300	3.047	2.215	7.978***	8.211***	7.790***
	(0.300)	(0.468)	(0.161)	(0.348)	(0.384)	(0.535)	(0.000)	(0.000)	(0.000)
Vol	-19.365***	-16.473***	-19.237***	-16.372***	-18.977***	-16.229***	3.684	3.916	3.640
	(0.004)	(0.008)	(0.004)	(0.009)	(0.004)	(0.009)	(0.458)	(0.427)	(0.463)
Lev	-0.000	0.022	-0.004	0.020	0.001	0.022	0.002	0.002	0.002
	(0.996)	(0.306)	(0.888)	(0.341)	(0.976)	(0.315)	(0.800)	(0.838)	(0.808)
Big4	5.758***	5.837***	5.807***	5.860***	5.770***	5.867***	2.722***	2.727***	2.703***
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.001)	(0.000)
Insratio	0.019**	0.018**	0.019**	0.018**	0.019**	0.018**	0.019	0.020*	0.019
	(0.011)	(0.029)	(0.012)	(0.030)	(0.011)	(0.029)	(0.104)	(0.094)	(0.108)
Ind FE	Yes	Yes	Yes	Yes	Yes	Yes	No	No	No
City FE	No	Yes	No	Yes	No	Yes	No	No	No
Year FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Firm FE	No	No	No	No	No	No	Yes	Yes	Yes
R <sup>2</sup> Adj	0.453	0.530	0.451	0.529	0.454	0.530			
R <sup>2</sup> Within							0.558	0.558	0.558
Obs.	3.688	3.685	3.688	3.685	3.688	3.685	3.688	3.688	3.688

Table 4. Baseline results. Impact of ESG-related visits on firm ESG disclosure score in the next year.

Notes: The dependent variable is *ESGD\_lead*, a firm's ESG disclosure score in the next year. Panel A reports the results adopting the pooled OLS regressions, and Panel B reports the results adopting panel regressions with firm-fixed effects. ESGvisit is a dummy indicating that a firm is visited and ESG issues are discussed during at least one visit in a calendar year. NESGvisit is a dummy indicating that a firm is visited and ESG issues are not discussed in any visits in a calendar year. Size is the market capitalization of a firm at the end of a calendar year. Vol is a firm's average quarterly volatility of stock returns in a calendar year where quarterly volatility is calculated as the standard deviation of daily returns across a calendar quarter. ROA is a firms' annual return on assets calculated as the average ROA across the quarters in a calendar year. Lev is a firm's leverage ratio calculated as the ratio of debt to equity at the end of a calendar year. Big4 is a dummy indicating that a firm has Big 4 auditors in a given year. Instatio is the percentage of institutional ownership in the firm at the end of a year. For specifications of Panel A, standard errors are clustered at firm level. For specifications of Panel B, Driscoll-Kraay standard errors are estimated. P-values reported in brackets. \*\*\*, \*\*, \* are 1, 5 and 10 percent statistical significance respectively.

	Panel A. ESC	3-related visit	Panel B. Non vis	-ESG-related
	(i)	(ii)	(i)	(ii)
Treat <sub>ESG</sub> x Post	0.271**	0.713**		
	(0.043)	(0.033)		
Treat <sub>NESG</sub> x Post			0.100	-0.104
			(0.423)	(0.574)
Size	32.802***	28.235***	31.904***	57.510***
	(0.000)	(0.000)	(0.000)	(0.000)
ROA	8.242***	5.269	10.496***	6.960**
	(0.000)	(0.112)	(0.001)	(0.024)
Vol	5.490	1.885	2.131	-1.337
	(0.275)	(0.633)	(0.548)	(0.789)
Lev	0.001	-0.289***	0.014	0.054**
	(0.931)	(0.000)	(0.494)	(0.034)
Big4	1.971*	3.209*	2.151**	0.763
	(0.075)	(0.051)	(0.040)	(0.214)
Insratio	0.009	0.014	0.012	0.012
	(0.447)	(0.383)	(0.352)	(0.170)
Firm FE	Yes	Yes	Yes	Yes
Year FE	Yes	Yes	Yes	Yes
R <sup>2</sup> Within	0.605	0.593	0.575	0.604
Obs.	2,534	1,440	2,343	1,236

Table 5. DiD estimation. Impact of ESG-related visits (Panel A) and non-ESG-related visits (Panel B) on firm ESG disclosure score in the next year.

Notes: The dependent variable is *ESGD\_lead*, a firm's ESG disclosure score in the next year. Specification (ii) of Panel A and B excludes the firms that have host non-ESG-related visits before hosting ESG-related visits, and the firms that have host ESG-related visits before hosting non-ESG-related visits, respectively. Treat<sub>ESG</sub> is a dummy indicating that firms ever host ESG-related visits during the sample period. Treat<sub>NESG</sub> is a dummy indicating that firms ever host non-ESG-related visits during the sample period. Post is a time indicator for post-treatment period. Size is the market capitalization of a firm at the end of a calendar year. Vol is a firm's average quarterly volatility of stock returns in a calendar year where quarterly volatility is calculated as the standard deviation of daily returns across a calendar quarter. ROA is a firms' annual return on assets calculated as the average ROA across the quarters in a calendar year. Lev is a firm's leverage ratio calculated as the ratio of debt to equity at the end of a calendar year. Big4 is a dummy indicating that a firm has Big 4 auditors in a given year. Driscoll-Kraay standard errors are estimated. P-values reported in brackets. \*\*\*, \*\*, \* are 1, 5 and 10 percent statistical significance respectively.

Table 6. Heckman t	wo-stage model.						
	Panel A. First-step Probit(ESCylisit)	(i)	(ii)	Panel B. S	econd-step	$(\mathbf{v})$	(11)
Ln(1+#firms)	0.129***	(1)	(11)	(111)	(IV)	(v)	(VI)
ESGvisit	(0.000)	0.797***	0.590**	$0.246^{***}$	$1.031^{***}$	$0.816^{**}$	$0.358^{***}$
NESGvisit		(0.003)	(0.010)	(0.001)	0.374	0.345 (0.295)	0.153
Size	-1.587 (0.238)	53.853*** (0.000)	50.136***	34.910*** (0.000)	53.541***	49.143***	34.725***
ROA	3.979***	1.828 (0.733)	3.259	6.032***	1.764 (0.742)	5.024	5.974***
Vol	0.644 (0.721)	-19.531***	-15.514**	3.798 (0.432)	-19.099***	$-14.970^{**}$	3.774 (0.434)
Lev	-0.022 (0.136)	0.012 (0.741)	0.015 (0.821)	0.013 (0.216)	(0.001) (0.781)	0.001	0.013 (0.242)
Big4	(0.120) (0.149) (0.324)	5.604***	5.849***	2.542***	5.637***	5.956***	2.529***
Insratio	(0.021) (0.000) (0.782)	0.018**	0.019**	0.018 (0.115)	0.019**	$(0.019^{**})$ (0.028)	0.018 (0.119)
Imr	(0.7.02)	-0.618 (0.639)	(0.344) (0.918)	-0.650** (0.048)	-0.488	(0.020) (1.092) (0.759)	$-0.623^{*}$ (0.058)
Ind FE	Yes	Yes	Yes	No	Yes	Yes	No
City FE	No	No	Yes	No	No	Yes	No
Year FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Firm FE	No	No	No	Yes	No	No	Yes
$R^2$ Pseudo $R^2$ Adj	0.073	0.451	0.528		0.451	0.528	
R <sup>2</sup> Within	3 664	3 664	3 661	0.558 3.664	3 664	3 661	0.558 3.664

Notes: The dependent variable of the specification in Panel A is a dummy indicating a firm is visited by institutional investors and ESG issues are discussed during at least one visit in a given year. The dependent variable of the specifications in Panel B is *ESGD lead*, a firm's ESG disclosure score in the next year. Ln(1+#firms) is the natural logarithm of one plus the number of firms visited in the city a given firm headquarters. ESGvisit is a dummy indicating that a firm is visited and ESG issues are discussed during at least one visit in a calendar year. NESGvisit is a dummy indicating that a firm is visited and ESG issues are discussed during at least one visit in a calendar year. NOI is a firm's average quarterly volatility of stock returns in a calendar year where quarterly volatility is calculated as the standard deviation of a firm at the end of a calendar quarter. ROA is a firm's annual return on assets calculated as the average ROA across the quarters in a calendar year. Lev is a firm's leverage ratio calculated as the ratio of debt to equity at the end of a calendar year. Big4 is a dummy indicating that a firm has Big 4 auditors in a given year. Instatio is the precentage of institutional ownership in the firm at the end of a vear. Imr is the inverse Mills ratio obtained from the first-stage regression. For Specifications (i)-(ii) and (iv)-(v) of Panel B, Standard errors are clustered at firm level. For Specifications (iii) and (vi) of Panel B, Driscoll-Kraay standard errors are estimated. P-values reported in brackets. \*\*\*, \*\*, \* are 1, 5 and 10 percent statistical significance respectively.

	Panel A. #	ESG-visits	Panel	B. #ESG-qs	Panel C	. Geodis	Panel D.	#UNPRI
	(i) Tert1	(ii) Tert3	(i) Tert1	(ii) Tert3	(i) Tert1	(ii) Tert3	(i) Tert1	(ii) Tert3
Treat <sub>ESG</sub> x Post	0.545	1.086***	0.385	0.740**	-0.115	1.032***	0.660*	3.116*
	(0.302)	(0.001)	(0.415)	(0.016)	(0.855)	(0.008)	(0.065)	(0.085)
Size	24.915***	92.773***	16.161***	72.001***	32.855***	26.469***	28.446***	-12.447
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.770)
ROA	10.922***	5.885	12.758***	-1.560	5.057	3.424	3.805	20.028
	(0.001)	(0.409)	(0.001)	(0.732)	(0.194)	(0.323)	(0.261)	(0.106)
Vol	9.144	12.612	8.529	37.505***	-4.961	12.320	0.193	4.613
	(0.296)	(0.256)	(0.289)	(0.007)	(0.219)	(0.372)	(0.954)	(0.521)
Lev	-0.270***	-0.060	-0.279**	0.417	0.101	-0.179	-0.365***	3.654
	(0.006)	(0.836)	(0.017)	(0.172)	(0.114)	(0.357)	(0.000)	(0.107)
Big4	0.765	5.222	0.167	8.100	3.485*	2.327	2.954*	-1.035
	(0.388)	(0.122)	(0.752)	(0.103)	(0.094)	(0.216)	(0.058)	(0.688)
Insratio	-0.006	0.026	0.001	0.037*	0.004	0.042**	0.011	0.180**
	(0.620)	(0.264)	(0.947)	(0.077)	(0.820)	(0.030)	(0.481)	(0.012)
Firm FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Year FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
R <sup>2</sup> Within	0.626	0.606	0.636	0.606	0.598	0.612	0.593	0.667
Obs.	830	459	731	430	574	596	1,358	132

Table 7. DiD estimation. Effects of ESG-related visits on firm ESG disclosure, by institutional investors' information demand.

Notes: Panels A-D report the results when the firms that host ESG-related visits are grouped by the tertiles of #ESG-visits, #ESG-qs, Geodis and #UNRPI. The dependent variable is ESGD\_lead, a firm's ESG disclosure score in the next year. Treat<sub>ESG</sub> is a dummy indicating that firms ever host ESG-related visits during the sample period. Post is a time indicator for post-treatment period. Size is the market capitalization of a firm at the end of a calendar year. Vol is a firm's average quarterly volatility of stock returns in a calendar year where quarterly volatility is calculated as the standard deviation of daily returns across a calendar quarter. ROA is a firms' annual return on assets calculated as the average ROA across the quarters in a calendar year. Lev is a firm's leverage ratio calculated as the ratio of debt to equity at the end of a calendar year. Big4 is a dummy indicating that a firm has Big 4 auditors in a given year. Instatio is the percentage of institutional ownership in the firm at the end of a year. Driscoll-Kraay standard errors are estimated. P-values reported in brackets. \*\*\*, \*\*, \* are 1, 5 and 10 percent statistical significance respectively.

	Panel A	. Not excluding	firms with	Panel B. Excluding firms with				
	nc	on-ESG-related	visits	non	-ESG-related vis	sits		
	(i)	(ii)	(iii)	(i)	(ii)	(iii)		
Treat <sub>E</sub> x Post	1.105**			1.196*				
	(0.024)			(0.064)				
Treats x Post		0.166			0.226			
		(0.508)			(0.288)			
Treat <sub>G</sub> x Post			-0.481			-0.130		
			(0.388)			(0.875)		
Size	65.569***	16.415***	7.453**	138.650***	6.155*	6.315		
	(0.000)	(0.000)	(0.043)	(0.000)	(0.056)	(0.209)		
ROA	14.989**	4.433	2.629	15.270*	5.363	-1.406		
	(0.020)	(0.330)	(0.591)	(0.080)	(0.227)	(0.889)		
Vol	11.383	-14.359	-2.740	16.531*	-35.098***	7.407		
	(0.242)	(0.184)	(0.722)	(0.059)	(0.000)	(0.528)		
Lev	0.011	-0.025	0.011	-0.759***	-0.077***	-0.618***		
	(0.705)	(0.189)	(0.481)	(0.002)	(0.006)	(0.003)		
Big4	6.645**	0.026	1.104	5.062*	2.909***	2.681		
	(0.029)	(0.984)	(0.282)	(0.085)	(0.000)	(0.149)		
Insratio	0.035	-0.006	-0.024	0.008	0.043**	-0.032*		
	(0.223)	(0.564)	(0.132)	(0.657)	(0.012)	(0.050)		
Firm FE	Yes	Yes	Yes	Yes	Yes	Yes		
Year FE	Yes	Yes	Yes	Yes	Yes	Yes		
R <sup>2</sup> Within	0.333	0.338	0.568	0.338	0.382	0.569		
Obs.	1,918	522	2,149	1,001	269	1,210		

Table 8. DiD estimation. Effects of ESG-related visits on firm ESG disclosure, by E-, S- and C	β-pillar.
--	-----------

*Notes*: Panels A and B include and exclude the firms that have hosted non-ESG-related visits prior to hosting ESG-related visits, respectively. The dependent variable in Specification (i) – (iii) is ED\_lead, SD\_lead and GD\_lead, respectively. Treat<sub>E</sub>, Treat<sub>S</sub> and Treat<sub>G</sub> are dummies indicating that a firm ever hosted E-/S-/G-related corporate visits, respectively. Post is a time indicator for post-treatment period. Size is the market capitalization of a firm at the end of a calendar year. Vol is a firm's average quarterly volatility of stock returns in a calendar year where quarterly volatility is calculated as the standard deviation of daily returns across a calendar quarter. ROA is a firm's annual return on assets calculated as the average ROA across the quarters in a calendar year. Lev is a firm's leverage ratio calculated as the ratio of debt to equity at the end of a calendar year. Big4 is a dummy indicating that a firm has Big 4 auditors in a given year. Instatio is the percentage of institutional ownership in the firm at the end of a year. Driscoll-Kraay standard errors are estimated. P-values reported in brackets. \*\*\*, \*\*, \* are 1, 5 and 10 percent statistical significance respectively.

Table 9. Did estimation. Effects of ESG felated visits on firm ESG disclosure, by proprietary cost.									
	Panel A. Not exe	cluding firms with	Panel B. Exc	Panel B. Excluding firms with					
	non-ESG-1	related visits	non-ESG	-related visits					
	(i) low HHI	(ii) High HHI	(i) low HHI	(ii) High HHI					
Treat <sub>ESG</sub> x Post	0.636***	-0.596	1.289*	0.161					
	(0.008)	(0.170)	(0.051)	(0.591)					
Size	30.568*	86.830***	25.646	103.616***					
	(0.094)	(0.000)	(0.150)	(0.000)					
ROA	13.043***	10.093**	11.407***	7.723					
	(0.000)	(0.032)	(0.000)	(0.359)					
Vol	12.969*	7.234*	10.961	4.785					
	(0.098)	(0.091)	(0.575)	(0.380)					
Lev	0.271***	0.003	0.166	-0.727***					
	(0.004)	(0.819)	(0.239)	(0.001)					
Big4	5.922**	0.488	6.565**	1.127					
-	(0.018)	(0.701)	(0.019)	(0.210)					
Insratio	0.002	-0.008	0.012	-0.009					
	(0.924)	(0.178)	(0.709)	(0.169)					
Firm FE	Yes	Yes	Yes	Yes					
Year FE	Yes	Yes	Yes	Yes					
R <sup>2</sup> Within	0.608	0.620	0.611	0.606					
Obs.	1,025	818	549	444					

Table 9. DiD estimation. Effects of ESG-related visits on firm ESG disclosure, by proprietary cost

*Notes*: Panels A and B include and exclude the firms that have hosted non-ESG-related visits prior to hosting ESG-related visits, respectively. Treat<sub>ESG</sub> is a dummy indicating that a firm ever hosted ESG-related corporate visits. Post is a time indicator for post-treatment period. Size is the market capitalization of a firm at the end of a calendar year. Vol is a firm's average quarterly volatility of stock returns in a calendar year where quarterly volatility is calculated as the standard deviation of daily returns across a calendar quarter. ROA is a firm's annual return on assets calculated as the average ROA across the quarters in a calendar year. Lev is a firm's leverage ratio calculated as the ratio of debt to equity at the end of a calendar year. Big4 is a dummy indicating that a firm has Big 4 auditors in a given year. Instatio is the percentage of institutional ownership in the firm at the end of a year. Driscoll-Kraay standard errors are estimated. P-values reported in brackets. \*\*\*, \*\*, \* are 1, 5 and 10 percent statistical significance respectively.

Table 10. Mechanishi unough uno	cats of exit. Impact of	corporate visits and LSO	disclosure score on fund	trading benaviors.
	(i)	(ii)	(iii)	(iv)
FSGD	0.011***	0 011***	0.012***	0.012***
LSCD	(0.000)	(0.000)	(0.001)	(0.001)
ESGvisit	0.251***	0.115*	(0.001)	(0.001)
	(0.000)	(0.087)		
ESGvisit x ESGD	()	0.004**		
		(0.028)		
NESGvisit			0.189***	0.128
			(0.000)	(0.219)
NESGvisit x ESGD				0.002
				(0.419)
Share	0.032*	0.032*	0.038*	0.037*
	(0.094)	(0.092)	(0.064)	(0.067)
ROA	-0.822**	-0.809**	-0.650**	-0.653**
	(0.015)	(0.016)	(0.042)	(0.042)
Vol	19.345***	19.366***	20.119***	20.113***
	(0.000)	(0.000)	(0.000)	(0.000)
Lev	-0.003	-0.003	-0.007	-0.007
	(0.876)	(0.879)	(0.653)	(0.652)
Size <sub>f</sub>	0.039***	0.039***	0.042***	0.042***
	(0.000)	(0.000)	(0.000)	(0.000)
Age <sub>f</sub>	0.530	0.530	0.512	0.512
	(0.189)	(0.188)	(0.228)	(0.228)
Fee	0.114***	$0.114^{***}$	0.116***	0.116***
	(0.000)	(0.000)	(0.000)	(0.000)
Flow	$0.108^{***}$	0.108***	0.108***	0.108***
	(0.001)	(0.001)	(0.001)	(0.001)
Size <sub>FC</sub>	0.232***	0.231***	0.238***	0.238***
	(0.000)	(0.000)	(0.000)	(0.000)
Num <sub>mng</sub>	-0.0/1*	-0.071*	-0.068	-0.068
	(0.053)	(0.055)	(0.106)	(0.106)
Gender	0.023	0.023	0.004	0.004
	(0.612)	(0.611)	(0.937)	(0.936)
Edu	0.035**	0.035**	0.030	0.030
	(0.043)	(0.044)	(0.103)	(0.104)
Num <sub>f</sub>	-0.002	-0.002	-0.004	-0.004
	(0.767)	(0.761)	(0.643)	(0.642)
Fund-firm pair FE	Yes	Yes	Yes	Yes
Year FE	Yes	Yes	Yes	Yes
R <sup>2</sup> Within	0.057	0.057	0.057	0.057
Obs.	293,902	293,902	306,982	306,982

Table 10. Mechanism through threats of exit. Impact of corporate visits and ESG disclosure score on fund trading behaviors.

Notes: The dependent variable is SHch, a fund's holding change of a portfolio firm, calculated as the number of shares of a portfolio firm a fund holds at the end of a given year minus the number of shares the fund holds at the end of last year. ESGD is a firm's ESG disclosure score in a given year. ESGvisit is a dummy indicating that a firm is visited and ESG issues are discussed during at least one visit in a calendar year. NESG visit is a dummy indicating that a firm is visited and ESG issues are not discussed in any visits in a calendar year. Share is the total number of shares outstanding at the end of a given year. Vol is a firm's average quarterly volatility of stock returns in a calendar year where quarterly volatility is calculated as the standard deviation of daily returns across a calendar quarter. ROA is a firms' annual return on assets calculated as the average ROA across the quarters in a calendar year. Lev is a firm's leverage ratio calculated as the ratio of debt to equity at the end of a calendar year. Sizef is a fund's total assets under management at the end of a year. Agef is the number of years since inception till current year. Fee is the fees charged. Flow is the quarterly fund flow in the year-end quarter. SizeFC is the total assets under management of the fund company the fund belongs to. Nummng is the number of managers managing a fund. Gender is the gender of a fund's manager(s) which is the average value of individual manager(s) gender, defined as one for male managers and zero for female managers. Edu is the education level of a fund's manager(s) which is the average value of individual manager(s)' education level, defined as one for high-school degree, two for bachelor degree, three for master degree, four for MBA/EMBA degree and five for PhD degree. Num, measures the busyness of a fund's manager(s), which is measured by the average number of funds within a fund-family that a fund's managers manage in the year-end quarter. Driscoll-Kraay standard errors are estimated. P-values reported in brackets. \*\*\*, \*\*, \* are 1, 5 and 10 percent statistical significance respectively.

	HZ_syn	HZ_Blbg	HZ_SP	Blbg_syn	SP_syn	SP_Blbg			
Panel A. Divergence measure in next year									
ESGD	-0.500***	-0.083	0.084	-0.749***	0.032	-0.042			
	(0.000)	(0.410)	(0.574)	(0.000)	(0.769)	(0.755)			
Size	-7.865	-29.866**	41.889**	17.014***	30.183	30.015			
	(0.485)	(0.012)	(0.015)	(0.003)	(0.299)	(0.372)			
ROA	30.490***	35.770*	12.268	17.501**	0.243	-38.290			
	(0.000)	(0.094)	(0.473)	(0.031)	(0.989)	(0.120)			
Vol	-46.012	-3.804	-21.611	-18.064	-174.804***	-0.847			
	(0.291)	(0.829)	(0.227)	(0.574)	(0.001)	(0.981)			
Lev	0.220***	0.692	0.086	-0.475	-0.075***	-2.730***			
	(0.005)	(0.263)	(0.455)	(0.719)	(0.003)	(0.006)			
Big4	2.427	-1.076	-2.867	1.887	-2.122	-2.716*			
	(0.697)	(0.738)	(0.248)	(0.113)	(0.387)	(0.053)			
Insratio	-0.056*	0.059	-0.048	0.081	0.119*	0.016			
	(0.065)	(0.422)	(0.134)	(0.180)	(0.062)	(0.818)			
Firm FE	Yes	Yes	Yes	Yes	Yes	Yes			
Year FE	Yes	Yes	Yes	Yes	Yes	Yes			
R <sup>2</sup> Within	0.026	0.016	0.027	0.047	0.025	0.103			
Obs.	1,443	836	1,307	647	869	651			
Panel B. Dive	ergence measure	e in two years							
ESGD	-0.509***	-0.165	-0.155	-0.720***	-0.413*	-0.327***			
	(0.000)	(0.204)	(0.136)	(0.000)	(0.052)	(0.000)			
Size	0.877	-26.359***	16.833	11.193	-42.224**	14.725*			
	(0.930)	(0.002)	(0.536)	(0.302)	(0.019)	(0.075)			
ROA	29.085	-2.357	62.440***	24.052***	24.265***	-8.304			
	(0.132)	(0.919)	(0.003)	(0.002)	(0.000)	(0.350)			
Vol	-81.204***	21.514	-8.293	34.578	-51.127*	-43.825			
	(0.002)	(0.553)	(0.455)	(0.121)	(0.094)	(0.163)			
Lev	0.064	0.819	0.037	1.154*	-0.089**	-1.404			
	(0.241)	(0.298)	(0.846)	(0.094)	(0.048)	(0.130)			
Big4	-7.584**	-7.910***	-6.019**	3.914***	3.821*	-0.082			
	(0.047)	(0.000)	(0.039)	(0.000)	(0.097)	(0.948)			
Insratio	-0.030	0.180***	0.046	0.062	0.001	-0.015			
	(0.584)	(0.001)	(0.373)	(0.192)	(0.967)	(0.800)			
Firm FE	Yes	Yes	Yes	Yes	Yes	Yes			
Year FE	Yes	Yes	Yes	Yes	Yes	Yes			
R <sup>2</sup> Within	0.029	0.032	0.038	0.044	0.022	0.097			
Obs.	1,413	786	1,184	639	849	617			

Table 11. Impact of ESG disclosure on ESG rating divergence.

*Notes*: The column names indicate the ESG rating divergence measures assessed in the specifications. Panel A reports the results when the value of the divergence measure in the next year is the dependent variable, and Panel B reports the results when the value of the divergence measure in two years is the dependent variable. ESGD is a firm's ESG disclosure score in a given year. Size is the market capitalization of a firm at the end of a calendar year. Vol is a firm's average quarterly volatility of stock returns in a calendar year where quarterly volatility is calculated as the standard deviation of daily returns across a calendar quarter. ROA is a firm's leverage ratio calculated as the ratio of debt to equity at the end of a calendar year. Big4 is a dummy indicating that a firm has Big 4 auditors in a given year. Instatio is the percentage of institutional ownership in the firm at the end of a year. Driscoll-Kraay standard errors are estimated. P-values reported in brackets. \*\*\*, \*\*, \* are 1, 5 and 10 percent statistical significance respectively.

# Appendix

Table A.1. Variable definitions

Variable	Definition	Data source						
Firm-level variables								
ESGD	ESG disclosure score for a calendar year	Bloomberg						
ED	Environmental-pillar disclosure score for a calendar year	Bloomberg						
SD	Social-pillar disclosure score for a calendar year	Bloomberg						
GD	Governance-pillar disclosure score for a calendar year	Bloomberg						
ESGVisit	A dummy indicating that a firm is visited and ESG issues are discussed during at least one visit in a calendar year	CSMAR and identification	manual					
NESGVisit	A dummy indicating that a firm is visited and ESG issues are not discussed in any visits in a calendar year.	CSMAR and identification	manual					
#ESG-visits	The total number of ESG-related visits a firm hosts in a given year	CSMAR and identification	manual					
#ESG-qs	The total number of ESG-related questions a firm receives in a given year	CSMAR and identification	manual					
Geodis	The median geographical distance across all the ESG-related visits a firm hosts in a given year.	CSMAR and identification	manual					
#UNPRI	The total number of UNPRI-signatory visitors a firm receives across all the ESG-related visits the firm hosts in a given year	CSMAR and website	UNPRI					
HHI	Herfindahl-Hirschman Index measure based on firm annual revenues for the industry a given firm belongs to	CSMAR						
Size	The market capitalization of a firm at the end of a calendar year.	CSMAR						
Vol	Average quarterly volatility of stock returns in a calendar year where quarterly volatility is calculated as the standard deviation of daily returns across a calendar quarter.	CSMAR						
ROA	Annual return on assets calculated as the average ROA across the quarters in a calendar year.	CSMAR						
Lev	Leverage ratio calculated as the ratio of debt to equity at the end of a calendar year.	CSMAR						
Big4	A dummy indicating that a firm has Big 4 auditors in a given year.	CSMAR						
Insratio	The percentage of institutional ownership in the firm at the end of a year.	CSMAR						
Share	The total number of shares outstanding at the end of year	CSMAR						
HZ_syn	The square of the difference in percentile ranking based on the ESG rating provided by Huazheng and Syntao Green Finance	WIND						

HZ_Blbg	The square of the difference in percentile ranking based on the ESG rating provided by Huazheng and Bloomberg	WIND and Bloomberg
HZ_SP	The square of the difference in percentile ranking based on the ESG rating provided by Huazheng and S&P Global	WIND and Bloomberg
Blbg_syn	The square of the difference in percentile ranking based on the ESG rating provided by Bloomberg and Syntao Green Finance	WIND and Bloomberg
SP_syn	The square of the difference in percentile ranking based on the ESG rating provided by S&P Global and Syntao Green Finance	WIND and Bloomberg
SP_Blbg	The square of the difference in percentile ranking based on the ESG rating provided by S&P Global and Bloomberg	WIND and Bloomberg
Fund-level v	ariables	
Size <sub>f</sub>	A fund's total assets under management at the end of a year	CSMAR
Age <sub>f</sub>	Number of years since a fund's inception till current year	CSMAR
Fee	Total fees (including management fees, distribution fees, subscription and redemption fees) of a fund at the end of a year	CSMAR
Flow	Quarterly fund flow in the year-end quarter	CSMAR
Size <sub>FC</sub>	Total assets under management of the fund company a fund belongs to at the end of a year	CSMAR
Num <sub>mng</sub>	The number of managers managing a fund at the end of a year	CSMAR
Gender	The gender of a fund's manager(s), which is the average value of individual manager(s)' gender, defined as one for male managers and zero for female managers	CSMAR
Edu	The education level of a fund's manager(s) which is the average value of individual manager(s)' education level, defined as one for high-school degree, two for bachelor degree, three for master degree, four for MBA/EMBA degree and five for PhD degree	CSMAR
Num <sub>f</sub>	The busyness of a fund's manager(s), which is measured by the average number of funds within a fund-family that a fund's managers manage in the year-end quarter	CSMAR

	i C ,
Type of investor relationship activities	✓ Visit by special entities □ Analysts meeting □ Media interview □ Performance announcement meeting □ News release □ Roadshow □ Site tour □ Other (please describe it here)
Participants	All Pensions Group (APG)
Time	March 19th 2020.
Location	Shenzhen
Insider attendees	Xuguang Lv, and board office, Ping An Bank Co., Ltd.
Content	

Table A.2. Disclosure of investor relationship records of Ping An Bank Co., Ltd..

**Question:** What social responsibilities does Ping An Bank undertake during the epidemic? Answer: During the outbreak of the novel coronavirus pneumonia in 2020, the Bank actively undertook its social responsibility in various ways and urgently carried out a series of public welfare activities as an active support for containing the spreading of the epidemic. (1) A first donation of RMB 30 million to Hubei Charity Federation was made in support of the anti-epidemic action and for the procurement of anti-epidemic supplies, insurances for volunteers, protective tools, supplements for community public health utilities, and so on. In addition, policies were specially made to give relief to affected customers such as medical workers engaged in the anti-epidemic action, customers infected with the novel coronavirus pneumonia, and customers in Hubei who were unable to make repayments as a result of the prevention and control measures. Those customers were allowed to make repayments later or pay less interest without leaving any bad credit records. ...(3) Databased loans assisted small- and medium-sized pharmacies in online financing. For medium-, small- or even mini-sized pharmacies, we provided industry-wide integrated and comprehensive online financial service plans. The data-based financing services specific for pharmacies are upgraded all around with "industry-specific data + online operations + intelligent risk control" to meet the needs of pharmacy customers, strengthen the online services for enterprises in the pharmaceutical industry, improve the convenient service level and support the real economy.

#### **Question**: What is the corporate strategy of Ping An Bank?

Answer: The Bank resolutely promotes the transformation of corporate business, focusing on refining and strengthening the three business pillars of "industry banking, transaction banking and comprehensive finance", promotes characteristic operation and ecological layout, and creates the "engine" of Ping An Group's comprehensive finance "1 + N" business; the Bank adheres to the "customer-centric" principle, focusing on the two core customer groups of "strategic customer group and small and micro customer group", continuously increasing the support for private enterprises, and fully satisfying the diversified needs of customers through data-based operation and differentiated services; firmly adhere to the "one lifeline" of asset quality without wavering and insist on a sustainable and high-quality development path, and provides solid support for building ""domestic best performer and world-leading intelligent retail Bank".

Attachments (If<br/>any)Slides for annual performance of 2019DateMarch 19, 2020.

	8	
Environment	Social	Governance
Beautiful China	child labour	assessment mechanism
carbon neutrality	collective bargaining	auditing committee
carbon reduction	community	auditing independence
contamination	discrimination	average pay
emission	donation	board
energy	employee turnover	bribery
energy conservation	employee welfare	business ethics
environment	enterprise foundation	corporate governance
environmental protection	equal pay for equal work	corruption
exhaust gas	fair trade	decentralization
green	female worker	diversity
greenhouse gas	forced labour	employee stock ownership
low carbon	freedom of association	incentives
peak carbon dioxide	human rights	independence
recycling	lost in accidents	independent audit
solid waste	non-regular employee	independent nomination
sustainable development	occupational fatality	information disclosure
utilization ratio	occupational health	internal governance
waste residue	occupational safety	management team reorganization
wastewater	professional training	managerial ownership
water conservation	public benefits	managerial power
water recycling	responsible	organization structure
	social responsibility	ownership structure
	supply chain management	remuneration
	supply chain supervision	salary gap
	vocational training	sustainability
	work-related injury	tax payment
		thematic committee
		unqualified opinion

Table A.3. Environmental, social and governance issue keywords.

*Notes:* Phrases that contain 'environment' but deliver other meanings are excluded from the list of keywords, e.g. economic environment, policy environment, market environment, competition environment, financing environment, development environment, new environment, regulation environment, domestic/international/foreign environment, operational environment, and information environment etc.

# **Internet Appendix**

То

# Institutional investors' behind-the-scene monitoring and ESG disclosure

Internet Appendix A: Description of the mutual fund subsample and ESG score data

Internet Appendix B: Supporting figures and tables

Figure IB: K density plots of the covariates for NNM

Table IB.1: Heckman two-stage model with alternative instrumental variable

Table IB.2: Effects of ESG-related visits on firm ESG disclosure, by information demand

Table IB.3: Summary statistics of ESG rating divergence

# **Internet Appendix A. Description of the mutual fund subsample and ESG score data** DESCRIPTION OF MUTUAL FUND SAMPLE

I obtain the data for actively managed equity open-ended funds from CSMAR. Similar to the existing literature (e.g., Chen et al., 2018; Chua and Tam, 2020; Agarwal et al., 2023), I exclude passively managed funds, bond funds, money market funds, and Qualified Domestic Institution Investor (QDII) funds from the sample, and define actively managed equity funds as the funds holding at least 50 percent of their assets in domestic stocks, including stock and mixed funds. Only primary share classes of funds are included. The screening gives me 2,758 mutual funds coming from 131 fund companies during the period of 2013-2020. After merging mutual fund data with the corporate visit data by visiting institution names, statistics show that the 131 fund companies visit 1,597 firms during the sample period.<sup>15</sup> For these funds, I collect information from CSMAR about funds' inception date, quarterly total assets under management (AUM), quarterly fees including management fee, distribution fee, custodian fee, subscription fee and redemption fee, manager characteristics including manager gender and education level, and semi-annual stock holdings.

#### DESCRIPTION OF ESG SCORE DATA

Huazheng started to release ESG rating for Chinese listed firms since 2009 which gives the longest time-series of ESG rating and covers the largest number of Chinese listed firms in comparison with other agencies. Huazheng rating classifies firms into eight grades ranging from C, CC, CCC, B, BB, BBB, A to AA according to ESG practices during the sample period. I define a firm's Huazheng ESG score as the value of the grade rated in which one to eight represents the eight grades from C to AA, respectively. Syntao Green Finance is a leading responsible investment

<sup>&</sup>lt;sup>15</sup> Corporate visit records are at fund company level rather than fund level. In most cases, it is impossible to know which specific fund within a fund company initiates the corporate visits. Thus, it is assumed that the funds within a same fund company share the information acquired from corporate visits, in line with Liu et al. (2017). The assumption is plausible given the evidence that funds within a fund company have an information-sharing channel (e.g. Gaspar et al., 2006; Elton et al., 2007).

professional service institution in China, grouping firms into ten grades between D and A+ based on the ESG practices since 2015. I define a firm's Syntao ESG score as the value of the grade rated in which one to ten represents the ten grades from D to A+, respectively. Bloomberg and S&P both started to rate Chinese firms' ESG performance since 2015, with Bloomberg (S&P) ESG score for the sample firms ranging from 0.72 to 5.58 (from 0 to 94).

# References:

Agarwal, S., Sulaeman, J., Zhang, Y. and Zong, Z., 2023. Does the Commitment to Responsible Investing Alter Investment Activities? Available at SSRN 4568714.

Chen, R., Gao, Z., Zhang, X., Zhu, M., 2018. Mutual fund managers' prior work experience and their investment skill. Financial Management 47(1), 3-24.

Chua, A. K. P., Tam, O. K., 2020. The shrouded business of style drift in active mutual funds. Journal of Corporate Finance 64, 101667.

Elton, E. J., Gruber, M. J., Green, T. C., 2007. The impact of mutual fund family membership on investor risk. Journal of Financial and Quantitative Analysis 42(2), 257-277.

Gaspar, J. M., Massa, M., Matos, P., 2006. Favoritism in Mutual Fund Families? Evidence on Strategic Cross-fund Subsidization. Journal of Finance 61(1), 73–104.

Liu, S., Dai, Y., Kong, D., 2017. Does it pay to communicate with firms? Evidence from firm site visits of mutual funds. Journal of Business Finance and Accounting 44(5-6), 611-645.

# Internet Appendix B. Supporting figures and tables

Figure IB. K density plots of the matching covariates for the treated and control groups of firms before and after NNM, respectively.

Panel A. Plots for ESGD



Panel B. Plots for *Ln(Size)* 



	Panel A. First-step			Panel B. Second-step				
	Probit(ESGvisit)	(i)	(ii)	(iii)	(iv)	(v)	(vi)	
Ln(1+#firms_esg)	0.197***							
	(0.000)							
ESGvisit		0.791***	0.625**	0.228***	1.029***	0.836**	0.346***	
		(0.004)	(0.016)	(0.001)	(0.006)	(0.021)	(0.005)	
NESGvisit		. ,	. ,		0.383	0.328	0.163	
					(0.243)	(0.297)	(0.123)	
Size	-1.648	53.622***	48.729***	34.958***	53.386***	48.652***	34.802***	
	(0.233)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	
ROA	3.894***	2.545	6.681	6.044***	2.223	6.141	5.880***	
	(0.000)	(0.574)	(0.319)	(0.001)	(0.624)	(0.356)	(0.001)	
Vol	0.357	-19.360***	-15.204**	3.899	-18.980***	-15.016**	3.856	
	(0.844)	(0.004)	(0.013)	(0.425)	(0.004)	(0.014)	(0.429)	
Lev	-0.021	0.007	-0.006	0.013	0.007	-0.005	0.013	
	(0.151)	(0.825)	(0.880)	(0.193)	(0.827)	(0.903)	(0.193)	
Big4	0.122	5.642***	5.945***	2.557***	5.663***	5.965***	2.540***	
e	(0.421)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	
Insratio	0.000	0.019**	0.019**	0.018	0.019**	0.019**	0.018	
	(0.797)	(0.011)	(0.025)	(0.114)	(0.011)	(0.026)	(0.118)	
Imr		-0.374	1.595	-0.651**	-0.327	1.513	-0.658**	
		(0.686)	(0.413)	(0.018)	(0.725)	(0.433)	(0.019)	
Ind FE	Yes	Yes	Yes	No	Yes	Yes	No	
City FE	No	No	Yes	No	No	Yes	No	
Year FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	
Firm FE	No	No	No	Yes	No	No	Yes	
R <sup>2</sup> Pseudo	0.086							
R <sup>2</sup> Adj		0.451	0.528		0.451	0.528		
R <sup>2</sup> Within				0.558			0.558	
Obs.	3,664	3,664	3,661	3,664	3,664	3,661	3,664	

Table IB.1. Heckman two-stage model, analogous to Table 6 with alternative instrumental variable.

Notes: The dependent variable of the specification in Panel A is a dummy indicating a firm is visited by institutional investors and ESG issues are discussed during at least one visit in a given year. The dependent variable of the specifications in Panel B is *ESGD lead*, a firm's ESG disclosure score in the next year. Ln(1+#firms\_esg) is the natural logarithm of one plus the number of firms hosting ESG-related visits in the city a given firm headquarters. ESGvisit is a dummy indicating that a firm is visited and ESG issues are discussed during at least one visit in a calendar year. NESGvisit is a dummy indicating that a firm is visited and ESG issues are not discussed in any visits in a calendar year. Size is the market capitalization of a visited firm at the end of a calendar year. Vol is a firm's average quarterly volatility of stock returns in a calendar year where quarterly volatility is calculated as the standard deviation of daily returns across a calendar quarter. ROA is a firms' annual return on assets calculated as the average ROA across the quarters in a calendar year. Lev is a firm's leverage ratio calculated as the ratio of debt to equity at the end of a calendar year. Big4 is a dummy indicating that a firm has Big 4 auditors in a given year. Instatio is the percentage of institutional ownership in the firm at the end of a year. Imr is the inverse Mills ratio obtained from the first-stage regression. P-values reported in brackets. \*\*\*, \*\*, \* are 1, 5 and 10 percent statistical significance respectively.

	Panel A. #	ESG-visits	Panel I	Panel B. #ESG-qs		Panel C. Geodis			Panel D. #UNPRI	
	(i) Tert1	(ii) Tert3	(i) Tert1	(ii) Tert3		(i) Tert1	(ii) Tert3		(i) Tert1	(ii) Tert3
Treat <sub>ESG</sub> x Post	0.122	0.845***	0.032	0.432*		-0.041	0.230	_	0.328***	1.250
	(0.647)	(0.000)	(0.877)	(0.057)		(0.911)	(0.493)		(0.007)	(0.174)
Size	26.048***	61.229***	18.757***	59.908***		29.186***	30.065***		32.363***	21.023***
	(0.000)	(0.000)	(0.000)	(0.000)		(0.000)	(0.000)		(0.000)	(0.001)
ROA	9.998***	5.673	10.597***	2.213		3.646**	7.018**		7.314***	12.065***
	(0.000)	(0.137)	(0.000)	(0.407)		(0.042)	(0.018)		(0.000)	(0.004)
Vol	14.738**	6.795	17.156**	2.714		-1.645	23.662		3.095	10.285
	(0.046)	(0.219)	(0.039)	(0.567)		(0.687)	(0.121)		(0.477)	(0.325)
Lev	-0.225***	0.014	-0.248***	0.014		0.015	-0.084		0.002	-0.046
	(0.000)	(0.120)	(0.000)	(0.185)		(0.615)	(0.274)		(0.849)	(0.158)
Big4	1.418*	1.799	1.294	1.373		2.689	3.172**		2.241*	-0.274
	(0.096)	(0.370)	(0.152)	(0.536)		(0.163)	(0.044)		(0.086)	(0.801)
Insratio	0.005	0.029**	0.009	$0.044^{***}$		0.008	0.009		0.010	0.023*
	(0.645)	(0.023)	(0.548)	(0.001)		(0.589)	(0.383)		(0.362)	(0.100)
Firm FE	Yes	Yes	Yes	Yes		Yes	Yes		Yes	Yes
Year FE	Yes	Yes	Yes	Yes		Yes	Yes		Yes	Yes
R <sup>2</sup> Within	0.616	0.601	0.620	0.605		0.597	0.608		0.601	0.644
Obs.	1,474	796	1,323	704		1,004	1,071		2,334	368

Table IB.2. Effects of ESG-related visits on firm ESG disclosure, by institutional investors' information demand. Analogous to Table 7 but the firms that have hosted non-ESG-related visits before hosting ESG-related visits are not excluded from the regressions.

Notes: Panels A-D report the results when the firms that host ESG-related visits are grouped by the tertiles of #ESG-visits, #ESG-qs, Geodis and #UNRPI. The dependent variable is ESGD\_lead, a firm's ESG disclosure score in the next year. Treat<sub>ESG</sub> is a dummy indicating that firms ever host ESG-related visits during the sample period. Post is a time indicator for post-treatment period. Size is the market capitalization of a visited firm at the end of a calendar year. Vol is a firm's average quarterly volatility of stock returns in a calendar year where quarterly volatility is calculated as the standard deviation of daily returns across a calendar quarter. ROA is a firms' annual return on assets calculated as the average ROA across the quarters in a calendar year. Lev is a firm's leverage ratio calculated as the ratio of debt to equity at the end of a calendar year. Big4 is a dummy indicating that a firm has Big 4 auditors in a given year. Instatio is the percentage of institutional ownership in the firm at the end of a year. Driscoll-Kraay standard errors are estimated. P-values reported in brackets. \*\*\*, \*\*, \* are 1, 5 and 10 percent statistical significance respectively.

Panel A. Correlations between ESG scores									
	Huazheng	Syntao	Bloomberg	SP					
Huazheng	1								
Syntao	0.275***	1							
Bloomberg	0.243***	0.412***	1						
SP	-0.013	0.175***	0.274***	1					
Panel B. Descrip	ptive statistics of E	SG rating diverge	ences						
	Mean	Median	Std Dev	p10	p90	Obs			
HZ_syn	13.585	4.000	19.272	0.000	49.000	1,636			
HZ_Blbg	13.162	4.000	17.663	0.000	36.000	890			
HZ_SP	19.593	9.000	23.171	0.000	49.000	1,774			
Blbg_syn	13.725	4.000	18.637	0.000	36.000	651			
SP_syn	15.624	4.000	19.605	0.000	49.000	904			
SP_Blbg	15.441	9.000	19.798	0.000	49.000	692			

Table IB.3. Pairwise Pearson correlations between ESG ratings (Panel A), and descriptive statistics for ESG divergence measures (Panel B).